

NOTA DI LAVORO 87.2015

The Migration Response to Increasing Temperatures

Cristina Cattaneo, FEEM and CMCC Giovanni Peri, University of California, Davis

Climate Change and Sustainable Development Series Editor: Carlo Carraro

The Migration Response to Increasing Temperatures

By Cristina Cattaneo, FEEM and CMCC

Giovanni Peri, University of California, Davis

Summary

Climate change, especially the warming trend experienced by several countries, could affect agricultural productivity. As a consequence, rural incomes will change, and with them the incentives for people to remain in rural areas. Using data from 116 countries between 1960 and 2000, we analyze the effect of differential warming trends across countries on the probability of either migrating out of the country or from rural to urban areas. We find that higher temperatures increased migration rates to urban areas and other countries in middle income economies. In poor countries, higher temperatures reduced the probability of migration to cities or to other countries, consistent with the presence of severe liquidity constraints. In middle-income countries, migration represents an important margin of adjustment to global warming, potentially contributing to structural change and even increasing income per worker. Such a mechanism, however, does not seem to work in poor economies.

Keywords: Global Warming, Emigration, Rural-Urban Migration, Agricultural Productivity **JEL Classification**: F22, Q54, O13

The research leading to these results has received funding from the Italian Ministry of Education, University and Research and the Italian Ministry of Environment, Land and Sea under the GEMINA project.

Address for correspondence Cristina Cattaneo Fondazione Eni Enrico Mattei Corso Magenta 63 20123 Milano Italy E-mail: cristina.cattaneo@feem.it

The Migration Response to Increasing Temperatures

Cristina Cattaneo (FEEM and CMCC) Giovanni Peri (University of California, Davis)*

September 23, 2015

Abstract

Climate change, especially the warming trend experienced by several countries, could affect agricultural productivity. As a consequence, rural incomes will change, and with them the incentives for people to remain in rural areas. Using data from 116 countries between 1960 and 2000, we analyze the effect of differential warming trends across countries on the probability of either migrating out of the country or from rural to urban areas. We find that higher temperatures increased migration rates to urban areas and other countries in middle income economies. In poor countries, higher temperatures reduced the probability of migration to cities or to other countries, consistent with the presence of severe liquidity constraints. In middle-income countries, migration represents an important margin of adjustment to global warming, potentially contributing to structural change and even increasing income per worker. Such a mechanism, however, does not seem to work in poor economies.

Key Words: Global Warming, Emigration, Rural-Urban Migration, Agricultural Productivity.

JEL Codes: F22, Q54, O13.

^{*}Cristina Cattaneo (cristina.cattaneo@feem.it), Fondazione ENI Enrico Mattei and CMCC, Corso Magenta, 63, 20123 Milan. Giovanni Peri (gperi@ucdavis.edu) Department of Economics, UC Davis, One Shields Avenue, Davis, CA 95616, USA. The research leading to these results has received funding from the Italian Ministry of Education, University and Research and the Italian Ministry of Environment, Land and Sea under the GEMINA project.

1 Introduction

One of the long-run effects of rising average surface temperatures is the disruption of productivity in agriculture. The optimal yield of agricultural products has been adjusted to local temperature for centuries, hence, productivity decreases as temperatures increase beyond a countries historical average (IPCC, 2014; Dell et al., 2014; Cline, 2007). Agriculture is still a relevant source of income and employment in poor countries, especially in rural areas. One potentially important margin of adjustment to declining agricultural productivity in poor countries is migration from rural to urban areas, either within the home country or another. While some papers have begun to analyze how warming may affect income per person across countries over the long run (e.g. Dell et al 2012), and other studies have analyzed the connection between temperature/precipitation and human migrations in some specific countries (e.g. Bohra-Mishra et al., 2014; Dillon et al., 2011; Mueller et al., 2014; Gray and Mueller, 2012a), only very few studies look at the systematic long-run effect of temperature change on emigration and rural-to-urban migration in poor and middle-income countries in the world.¹ This paper gathers data and proposes a model and simple empirical framework to analyze the impact of temperature change on emigration rates in countries where agriculture is still an important sector and many migrants originate from rural areas.

By impoverishing the rural population of poor countries and worsening their income perspectives, long-term warming may affect migration in different ways, depending on the initial income of those rural populations. As previously suggested by studies such as Mayda (2010), a decline in the income of the sending country may have a depressing effect on the share of emigrants from very poor countries. In these countries, individuals are near subsistence, so a lower income worsens their liquidity constraint, implying potential migrants have a reduced ability to pay migration costs. In this case, global warming may trap very poor rural workers who become unable to leave agriculture, worsening their poverty. To the contrary, in countries in which individuals are not extremely poor, a decline in agricultural income may provide incentives to migrate to cities or abroad. Decreasing agricultural productivity may encourage a mechanism that ultimately leads to economic success for migrants, benefiting their country of origin and shifting people out of agriculture into urban environments. The inverted U-shape of migration rates as a function of income per person in the countries of origin is usually rationalized in this type of framework. However, we are not aware of a simple formalization of this model nor of a clean analysis which tests this non-monotonic

¹Cai et al. (2014) is probably the paper more closely related to ours. It analyzes specifically the link between temperature, crop yields and migration to OECD countries. They use, however, yearly data between 1980-2000 and only migration to OECD coutries, capturing therefore short-run relationships and long distance migration.

effect by exploiting variation of an exogenous determinant of income per person, such as temperature.

In this paper we use a simple framework that extends the classical Roy-Borjas (Roy, 1951; Borjas 1987) model and uses it to analyze the effects of exogenous changes in agricultural productivity (due to temperature increase) and its opposite effects on the probability of emigration for poor or middle-income countries. In particular, the model predicts that a long-run increase in temperature that decreases the income of rural populations in very poor countries generates a poverty trap and lower probability of emigration. To the contrary, for middle-income countries, the decline in agricultural productivity push emigration from rural areas. This stimulates urbanization and may speed the country's structural transformation, ultimately increasing its income per person. In accordance with the model's predictions, we find that in very poor countries increasing temperatures decrease rural emigration and urbanization, while in middle- income countries they increase urbanization and emigration. We then show long-run warming speeds the transition from agriculture to non-agriculture in middle-income countries. Conversely, it slows this transition in poor countries – worsening the poverty trap – as poor rural workers become less able to move to cities or abroad. We also find the emigration in middle-income countries that is induced by higher temperatures is local and is associated with a growth in GDP per person, while the decline in emigration and urbanization in poor countries is associated with lower average GDP per person.

The rest of the paper is organized as follows. Section 2 reviews the literature on climate and international migrations. Section 3 presents a simple variation of the Borjas-Roy model relating agricultural productivity to migration rates at different income levels. Section 4 describes the data and variables and section 5 presents the main empirical specifications and the main estimates of the effects of warming on migrations. Section 6 shows some robustness checks and section 7 checks that the connection climate-migration is consistent with the estimated effects of climate on structural change and GDP across countries. Section 8 concludes the paper.

2 Literature Review

The literature analyzing the effects of weather and climate events on migration is recent and growing fast. Several papers have analyzed the impact of episodes of drought, high temperature, or low precipitation on rural emigration in some specific countries. Dillon et al. (2011) analyze migration in Nigeria. Mueller et al. (2014) look at the connection between temperature variation and migration in Pakistan. Gray and Mueller (2012a) consider the link between draughts and emigration in Ethiopia, while Gray and Mueller (2012b) analyse

the effect of flood on mobility in Bangladesh. Gray and Bilsborrow (2013) and Gray (2009) analyze internal and international migration in Ecuador in response to rainfall. Henry et al. (2004) look at the case of annual precipitations and migration in Burkina Faso. Bohra-Mishra et al. (2014) analyze Indonesia and Kelley et al. (2015) focus on Syria. Because of its extreme poverty and dependence on agricultural production and employment, Sub-Saharan Africa has been a main area of attention. Most of these studies, however analyze the yearly correlation between weather phenomena and migration and may pick up temporary displacements rather than long-term trends. In multi-country studies of sub-Saharan Africa, Barrios et al. (2006) analyze the link between average rainfall and urbanization, and Marchiori et al. (2012) estimate how temperature and precipitation anomalies have affected migration in sub-Saharan Africa.

Another case that has been studied in depth is the connection between climate and migration from Mexico. Looking at Mexico-US migrations Munshi (2003) was the first to show the connection between low rainfall and migration rates from Mexico to the US. More recently, Feng et al. (2010) confirm the relation between weather and migration from Mexico. However, Auffhammer and Vincent (2012) demonstrate this effect vanishes after they control for a richer set of covariates. Overall, the existing literature on weather/climate change and migration focuses on within country data and usually on gross yearly migration rates. Hence it fails to provide a general picture on the potential long-run effect of weather changes on migration across countries. Some econometric analyses at the macro level exist, but they mainly focus on the consequences of natural disasters, such as droughts, earthquakes, floods, storms, and volcanic eruptions. They do not directly tackle the question regarding the effect of changes in average temperatures on migrations in the long-run. Beine and Parsons (2015) produced an accurate study that focuses on bilateral migration and analyzes the impact of extreme weather events, deviations and anomalies in temperatures from the long-run averages, after one controls for many other bilateral factors. The narrow focus on partial effects and on some extreme events makes that paper different from ours. Our paper differs from all the previous ones by considering all countries of the world and explicitly analyzing the effects of temperature on migration within a simple Roy-Borjas model of migration and average productivity. In so doing, it identifies a crucial distinction of temperature increases on poor and middle-income countries and tests whether such distinctions and other additional implications are supported by the data. Finally the paper closer to our approach is Cai et al (2014). In this paper the authors analyze how yearly bilateral migration flows depend on yearly temperatures at origin for a panel of 163 countries of origin into 42 OECD destinations for the period 1980-2010. The structure of the analysis implies that these are short-run elasticity responses (within the year) and only includes migration to OECD countries. The authors do not separate between poor and middle income countries and use quite noisy data on gross flows of migrants, instead of Census based data on net migrations. Significant short-run temporary migration can be captured by that design. We are more interested in the long-run impact of slow changing temperature and precipitation on migration rates.

3 A Simple Framework

3.1 The migration decision

Consider two countries defined as "Poor", P, and "Middle-Income", M, where workers, who are potential migrants to a third country "Rich" (R), live and work. We consider a very simple two-period model, in the spirit of Roy-Borjas (Roy, 1951; Borjas, 1987), that delivers a "hump-shaped" emigration rate as a function of the country of origin's income per person (consistently with the empirical literature from Zelinsky (1971) to Hatton and Williamson (1994, 2003 and 2011)). In the first period, individuals differ in their skills, work in their country of origin (P or M), and earn the local wage. At the beginning of the second period, individuals choose between migrating to country R or staying in their country, based on the comparison of their wage during the second period. If they stay in the country of origin they earn w_{iJ} . If they migrate to R they earn w_{iR} , but must pay up-front monetary and non-monetary migration costs. For simplicity we assume individuals have 0 discount rate, the wage in the country of origin for period 1 and 2 are identical, and no uncertainty exists. The wage of individual i when residing in country of origin J (= P, M) in the first and second period can be written as:

$$w_{iJ} = \mu_J + \beta_J \varepsilon_i \quad J = P, M \tag{1}$$

where μ_J is the basic income/wage in country J earned by a person with median skills. We can imagine this term depends positively on agricultural productivity – among other factors – especially as the economy of country P and M depend on agriculture and agriculturerelated sectors. Through agricultural productivity, therefore, the median income in country J depends on its temperature T_J , expressed as: $\mu_J(T_J)$. The term β_J represents the return to skills in country J. The term ε_i is a measure of skills of individual i that we assume, for simplicity, as normally distributed with an average of 0 and a standard deviation of 1. If the same individual were to migrate to country R he/she would earn the following wage instead:

$$w_{iR} = \mu_R + \beta_R \varepsilon_i \tag{2}$$

For simplicity we have assumed the skills of the individual, measured by ε_i , are perfectly transferable from P or M to country R. However, the returns to skills in country R are different than in the origin country. Following strong evidence from the existing literature (Grogger and Hanson (2011) and Ortega and Peri (2012)) we assume the rich country has higher median wage and higher skill premium than the poor and middle-income countries. Moreover, following most of the literature on climate change (Dell et al., 2014), we assume temperature changes have an effect on agricultural productivity (relevant for country J =P, M), but not (or much less) on non-agricultural productivity (relevant for country R) so that the dependence of μ_R on T_J can be ignored. These assumptions correspond to the following restrictions on the parameters: $\mu_R > \mu_J$ and $\beta_R > \beta_J$ for $J = P, M.^2$ For simplicity, we also assume the distribution of skills, ε_i , is identical in country P and M and the cost of migrating from either of them to R, the rich country, is equal and can be expressed as $(C_{Mon} + C_{Non})$ where C_{Mon} are monetary costs of migrating – such as cost of relocating, traveling, and searching – while C_{Non} are the non-monetary (psychological) costs. Both are expressed in units of labor compensation. We assume individual's have linear preferences in their net wages (wages net of migration costs), and within this very simple framework the decision to migrate for individual i implies a comparison between the net income when migrating and staying. Thus, the individual will migrate from country J to R if:

$$\mu_R + \beta_R \varepsilon_i - C_{Mon} - C_{Non} > \mu_J + \beta_J \varepsilon_i, \tag{3}$$

or more simply:

$$\varepsilon_i > \frac{\mu_J(T_J) - \mu_R + C_{Mon} + C_{Non}}{\beta_R - \beta_J}.$$
(4)

Condition (4) has been typically thought of in the literature as a "selection" equation. The parameter restriction $\beta_R > \beta_J$ implies "positive selection". Namely, as shown in equation (4), only individuals with skills above a certain level have incentives to migrate. This is consistent with abundant evidence as summarized, for instance, in Docquier et al. (2011). Alternatively, we can see equation (4) as an incentive-compatible constraint. Namely, individuals from country J will migrate only if their gains from migration (wages at destination) exceed the opportunity cost (wage at home) plus migration costs (monetary and non-monetary). The lower the threshold in (4), the larger is the share of individuals for which the incentive constraint is satisfied. The migration decision, however, should also

²Under these assumptions, and if costs of migration are equal between M and R, P and R, and P and M, we do not have to consider potential migration between P and M as workers from either country would want to migrate to R.

satisfy a "feasibility" constraint. If we assume migration takes place at the beginning of the second period and individuals in country P and M cannot borrow (liquidity constraint), then they can migrate only if the monetary costs of migration does not exceed their total savings at the end of period 1, which, in our simple model, is at most equal to w_{iJ} . With labor as the only source of income and assuming monetary costs of migrations must be paid up front, the necessary condition for feasibility – which can be called a liquidity constraint – can be written simply as:

$$\mu_J(T_J) + \beta_J \varepsilon_i > C_{Mon} \tag{5}$$

or:

$$\varepsilon_i > \frac{C_{Mon} - \mu_J(T_J)}{\beta_J} \tag{6}$$

3.2 Implications on Emigration Rates

Using the fact that individual skills ε_i are distributed in the population of country J as a normal with 0 mean and unitary variance, the two conditions (4) and (6) above imply the fraction of people who will migrate from country J is equal to one minus the cumulative density of a normal distribution at the threshold defined in (4) and (6). For each country only one of the two thresholds can be binding. It is easy to see that the "incentive" threshold (4) is increasing in the "median" income $\mu_J(T_J)$, while the "liquidity" threshold (6) is decreasing in it. The monotonicity of the two thresholds implies that there is a value of $\mu_J^*(T_J^*)$ for which they are identical and we consider that value as marking the divide between "Poor" (P) and "Middle income" (M) countries³. Hence, this model provides two very clear predictions:

Proposition 1 For Middle-Income Countries, an increase in average temperature is associated with an increase in the emigration rate.

Proof. For countries whose median income is higher than $\mu_J^*(T_J^*)$, defined as Middleincome countries, M, only the threshold (4) is binding. Hence the share of people migrating is the one with skills above that threshold, given by:

$$\frac{Mig_M}{Pop_M} = 1 - \Phi\left(\frac{\mu_J(T_J) - \mu_R + C_{Mon} + C_{Non}}{\beta_R - \beta_J}\right)$$
(7)

where Φ is the CDF of a standard normal distribution. The expression on the right hand side is decreasing in μ_J (because the CDF Φ is a monotonically increasing function). If we

³That value is defined as: $\mu_J^*(T_J^*) = \frac{(\beta_R - \beta_J)C_{Mon} + \beta_J(\mu_R - C_{Mon} - C_{Non})}{\beta_R}$

assume that increases in temperature T decrease basic agricultural productivity μ_J , then the expression is increasing in T_J .

The intuition is straightforward. As lower agricultural productivity implies lower median income, in middle-income countries this effect increases the incentive (and hence probability) of migrating and hence raises the emigration rate. For those countries the liquidity constraint does not bind.

Proposition 2 For Poor Countries an increase in average temperature is associated with a decrease in the emigration rate.

Proof. For countries whose median income is lower than $\mu_J^*(T_J^*)$, defined as Poor countries, P, differently than for the other group, only the liquidity threshold (6) is binding. Hence the share of people migrating is the one with skills above that threshold, given by:

$$\frac{Mig_P}{Pop_P} = 1 - \Phi\left(\frac{C_{Mon} - \mu_J(T_J)}{\beta_J}\right) \tag{8}$$

where Φ is the CDF of a standard normal distribution. The expression on the right hand side is now increasing in μ_J . If, as before, we assume that increases in temperature T decrease median productivity μ_J , then the expression above would be decreasing in T_J .

The intuition is also straightforward. In poor countries, the liquidity constraint is binding. Hence, lower agricultural productivity makes people poorer, decreasing their ability to pay migration costs, hence reducing the emigration rate. For these countries the incentive to migrate is very high, but individuals are simply too poor to afford migration, which is only worsened by lower agricultural productivity.

Figure 1 illustrates these two cases in Panels 1 and 2, respectively. Panel 1 represents the skill distribution in the middle-income country. We see the migrating population is the one in the shaded area with skills above ε_I (T_I), the skill-threshold determined by the incentive-constraint. On the contrary, the skill-threshold driven by the liquidity constraint, ε_L (T_L), is not binding and, hence, irrelevant. The arrows in the graph represent the shift of the thresholds implied by an increase in temperature, T. As a consequence of increases in temperature, the upper (incentive) threshold moves to the left, while the lower (liquidity) threshold – irrelevant for middle- income countries – moves to the right. This implies the area below the skill density distribution and to the right of the threshold increases. Panel 2 shows the picture for poor countries. We assume the same relative distribution of skills, but in this case the ordering of the thresholds is switched. The liquidity threshold that moves to the right as T increases is now binding. This implies a smaller mass of people migrating as a consequence of higher temperatures. By taking logarithms and log-linearizing both sides of each equation (7) and (8) and merging them into one equation, we obtain the basic equation and prediction for our empirical test and analysis. Namely, considering a generic country j that can be M (middle income) or P(poor) we can write:

$$\ln\left(\frac{Mig_j}{Pop_j}\right) = \alpha + \gamma \ln T_j + \gamma_P \ln T_j * D(j \in P) + \beta C_j$$
(9)

In (9) the dependent variable is the natural logarithm of the migration rates from country j and it depends on the logarithm of the average temperatures in the country, $\ln T_i$. To capture the different dependence in poor and middle-income countries, we allow for a linear term whose coefficient γ captures the effect of temperature on emigration rates in middle-income countries. We then add an interaction with the dummy $D(i \in P)$ that is equal to 1 if country j is a poor country, for which the "liquidity threshold" is binding and 0 otherwise. With this notation, the parameter γ captures the elasticity of emigration rates to average temperature for medium-income countries and $\gamma + \gamma_P$ captures the elasticity for poor countries. The term C_j captures potential determinants of migration costs in country of origin j. Let us also notice that if we interpret "R" as the urban areas, and M and P as the rural areas in the Middle-income or Poor country, the model above can be interpreted as a model of rural-to-urban migration. Even in that case, it makes sense that migration is skill-intensive and the incentive condition affects migration in middle-income countries, while the liquidity constraint affects it in poor countries. Hence the consequence of warming would be more urbanization in middle-income countries, but less urbanization in very poor countries. The prediction of the model can be summarized, within the compact format of expression (9) above, as follows:

- 1. As the average temperature of a middle-income country increases, reducing its agricultural productivity relative to urban productivity, we expect workers to migrate abroad and to the cities at higher rate. Therefore the model predicts $\gamma > 0$.
- 2. As the average temperature of a poor country increases, reducing its agricultural productivity, we expect workers whose average income is very low to have fewer resources to pay for their migration possibilities. Therefore the model predicts $\gamma + \gamma_P < 0$.

Our empirical analysis focuses on estimating the link between temperature and emigration, and will provide important evidence to evaluate the predictions of the model.

4 Data and Summary Statistics

In order to test the empirical predictions of the model, we merge data on the average temperature and on international migration and urbanization for all available countries in the world between 1960 and 2000.⁴ The data on temperatures are taken from Dell et al. (2012). In our empirical specifications we also control for a measure of annual precipitation, whose long-run behavior can affect agricultural productivity. This variable is used as a control in Dell et al (2012) because changes in precipitation can be an important aspect of longrun climate trends affecting agricultural productivity. Moreover, given that precipitation and temperature are historically correlated, both temperature and precipitation need to be included in the empirical specification to obtain unbiased coefficients (Auffhammer et al., 2013). The (terrestrial) monthly mean temperature and precipitation data at 0.5×0.5 degree resolution, obtained from weather stations (Matsuura and Willmott, 2007), are aggregated into country-year averages using the population in 1990 at 30 arc second resolution (CIESIN et al. 2004) as weights. In an alternative approach, used as a robustness check, the weather station data are averaged using area, rather than population, weights. In some specifications, in order to analyze whether long-term warming affects countries by increasing the probability of extreme weather events, we also include the incidence of droughts, floods, storms and extreme heat as controls. Those data are taken from the International Disaster Database compiled by the Centre for Research on the Epidemiology of Disasters (Guha-Sapir et al., 2015).

The migration data are taken from Ozden et al. (2011), and include bilateral migrant stocks between 116 countries in the last five available census years spanning the period from 1960 to 2000. Drawing from the bilateral data, we compute net emigration flows as differences between stocks in two consecutive censuses. We first sum all net flows for the same countries of origin and compute emigration rates as the ratio between the aggregate net flow of emigrants in the decade relative to the origin country population at the beginning of the decade.⁵ The data on urbanization rates are taken from the World Urbanization Prospects (UN, 2014). They measure the share of the population of a country living in urban areas between 1960 and 2000 available over ten year intervals. For GDP per capita the main sources are the Penn World Table (2009) and the World Development Indicators (World Bank, 2015). Finally, data on the value added in agriculture are from the World Development Indicators (World Bank, 2015).

⁴Further details on the data and the full list of countries classified as either "poor" or "middle income" can be found in the Data Appendix A.

 $^{{}^{5}}$ Bilateral net flows that are negative (usually very small numbers) are set to 0 as they may be due to mortality of the stock of emigrants abroad.

Consistently with our model, the set of countries of origin we consider for our analysis are those that can be considered "poor" or "middle income" according to their income per person. These are the countries for which temperature changes may have the largest productivity effect, because agriculture contributes a significant share of GDP. In practice we define poor and middle-income countries in two ways. In a first definition, we consider in our sample all non-OECD countries,⁶ which gives us a total of 115 countries. In a second definition we rank countries by PPP-adjusted per capita GDP in 1990, taken from the Penn World Table, and we choose those below the top quintile, which leaves us with 116 countries. In the first definition we consider poor countries those in the bottom quartile of the non-OECD sample income distribution, measured as PPP-adjusted per capita GDP in 1990. In the second definition, poor countries are those in the bottom quintile of the sample income distribution, computed before excluding rich countries. Under both definitions we end up with the same list of 30 poor countries, while the list of middle-income countries is somewhat different between the two definitions (see the Data Appendix for each list). Ideally, one would want to use 1970 as the reference year to partition countries between poor and middle income. but this choice would drastically reduce the sample of countries. Given the relative stability of country ranking in per capita GDP we are confident that our choice would mostly overlap with one based on the 1970 definition of GDP per person. The countries near the threshold between poor and middle income are those with yearly income per person around \$1,500 in 1990. This is clearly a low threshold, implying a large share of the "poor" countries are in sub-Saharan Africa. For rural population in these countries, which tends to be the poorest portion within the country, the liquidity constraints is clearly very relevant as they likely live on an income of a few dollars per day. Saving some hundreds of dollars needed to move out of the country can be very hard for these families. The threshold between middle-income and rich countries was instead around \$15,000 per person in 1990 which was about the income per person of Portugal or Greece. Rich countries are important destinations for migrants from poor and middle-income countries of origin, but they are not included in our analysis as sending countries.

Table 1 provides the summary statistics of the variables of interest for the two groups of countries (poor and middle income) separately when we include all non-OECD economies. Several features of the data are worth discussing. First, the average ten-year emigration rate for middle-income countries is 4.2%, including migration to both OECD and non-OECD destinations. This average is much higher than for poor countries, whose decennial net rate is 1.8%. This is consistent with the idea that emigration rate grows with income, up to a

⁶The Organization for Economic Cooperation and Development (OECD) is usually considered as the club of developed countries. It includes most of the countries in the world with high GDP per person.

certain level. Second, income per capita and urbanization rates are much higher in middleincome countries than in poor countries. In particular, the share of urban population is 42% in middle-income countries and only 19% in poor countries. Though, both are far from the level of urbanization in rich countries (around 75%). Additionally, a substantial share of value-added production in poor countries comes from agriculture, around 35%, and is a non-negligible source of GDP (accounting for about 16%) in medium-income countries, as well.

The differences in emigration rate and temperature trends are depicted in Figures 2 and 3. The graphs show the evolution of emigration rates and temperatures for ten selected poor and middle-income countries, chosen to be each at a decile of the overall distribution for the total four decade change. In each figure we standardize the average emigration rate and average temperature of each country in the first decade to zero, making even small variations apparent. The left-hand panel of Figure 2 shows that emigration rates are relatively stable between whole period in middle-income countries, with most countries experiencing changes of only few percentage points. Exceptions are Albania, whose emigration rate increased 28 percentage points, and Algeria, whose emigration rate decreased, especially between the first and second decade, by 9 percentage points. For poor countries, we can observe a larger proportion of increasing emigration rate than decreasing it, with a significant amount of variation.

As for temperature, Figure 3 shows that over the considered period temperatures increased in the large majority of middle-income and poor countries. As one can see from the figure, the last decade was generally warmer for all countries than the first; the temperature changes over the period are in fact positive with the exception of countries in the bottom decile of the distribution of temperature changes. We also observe significant variation in the amount of warming experienced over three decades, with a range of about 1 degree Celsius separating the top two deciles for both middle-income and poor countries.

Figure 4 shows our most interesting stylized fact. It plots long-term changes in temperature against long-term changes in emigration rates for poor and middle-income countries separately, along with a fitted regression line. In particular, we take the difference between the (natural log of) average temperatures and emigration rates in the first two decades (1960-80) and in the last two decade (1981-2000) of our data and plot one against the other. The difference in the relationship between the two groups of countries is clear. Middle-income countries show a positive (albeit not strong) correlation while poor countries show a negative these are the types of correlations predicted by our model. We will test the robustness of these correlations more systematically in the next section.

5 Empirical Specification and Results

Following specification (9) suggested by the model, we estimate the following empirical specification:

$$Y_{j,t} = \alpha + \gamma \ln(T_{j,t}) + \gamma_P \ln(T_{j,t}) * D_j + \delta \ln(P_{j,t}) + \delta_P \ln(P_{j,t}) * D_j + \phi_j + \phi_{r,t} + \phi_{p,t} + \epsilon_{j,t}$$
(10)

The variable $Y_{j,t}$ captures the outcome of interest in country j and in the decade beginning with year t = 1960, 1970, 1980, 1990. It will be alternatively the natural logarithm of the emigration rates (described in the previous section) or the average urbanization rate, computed as the urban population over the total population of country j during the decade beginning in t.⁷ $T_{j,t}$ represents the average temperature of country j during each decade beginning with year t and $P_{j,t}$ captures the ten year average precipitation. The inclusion of both temperature and precipitation in the estimated specification follows the literature that studies the effect of climate change on any outcome. Both the natural logarithm of the temperature and of the precipitation are entered linearly, as well as interacted with the dummy D_j that equals one if country j is categorized as "Poor". This allows different elasticity estimates for poor and middle-income countries, a point emphasized in the model. We also include country fixed effects, ϕ_i , capturing fixed country characteristics such as their geography and institutions. The term $\phi_{r,t}$ captures region-decade dummies in order to absorb regional factors of variation in economic conditions over time and $\phi_{p,t}$ are decade fixed effects interacted with a poor country dummy, to capture differential time variation in the group of countries considered as "poor" relative to those considered as "middle income". $\epsilon_{j,t}$ is a random error term that can have a correlation within country; hence our choice to cluster at the country level when estimating. As emphasized in the previous section, we only consider a sample of middle-income and poor countries of origin. In the main specification we apply the first definition of poor and middle-income countries and include only non-OECD countries of origin equating OECD to rich countries. Alternatively, in robustness checks we apply the second definition and consider as "rich" (and drop from the country of origin sample) those countries in the top quintile of the income per person distribution. The dummy for "poor" countries is defined as equal to one for countries in the bottom quartile of the sample income distribution in the non-OECD sample. It is equal to one for countries in the bottom quintile of the income distribution (determined before excluding rich countries) in the sample that excludes top income countries.

Specification (10) is based on the model presented in section (3). It also represents a

⁷For urbanization rates, our first decade starts in year 1950 as we have data going back to that date.

simple reduced-form linear relationship between temperature and migration allowing such a relation to vary depending on the initial income per person in the country of origin. While it is clear that average temperature is an exogenous variable, the real question is: through what channels does temperature operate on migration? In our model and analysis we focus on specific implications of a model in which the main channel operates through a decrease in agricultural productivity and rural income, both of which are not easily observable variables for our panel of countries. One option would be to include several controls such as population size, sociopolitical environment and others in the regression to reduce the scope of omitted channels. However, as those variables are themselves affected by temperature, including them may produce a bias in the estimation by introducing an over-controlling problem. The estimation of an equation that controls for both temperature and other variables that are influenced by the temperature would not capture the total net effect of temperature on migration (Dell et al., 2014). Therefore, we decided to remain consistent with the literature (Jones and Olken, 2010; Dell et al., 2012) by including only fixed effects as controls. We then directly analyze the potential channels by assessing the impact of temperature on income per person and agricultural value added as outcomes to see whether the estimated effects on those variables are consistent with the working of our model.

5.1 Effects on International Migration

The main estimated coefficients capturing the effects of average temperature on international migrations are presented in Table 2. Columns (1) to (4) and (7)-(8) show estimates in which we use the population weights for the aggregation of weather station temperature and precipitation data, while Columns (5) and (6) aggregate temperature data using area weights. In Columns (1) to (6) OECD countries are excluded, so the sample of poor and middleincome countries is defined as non-OECD ones. In Columns (7) and (8) countries in the top quintile of income per capita distribution are dropped in identifying poor and middle-income countries. The estimated specifications in Columns (2), (6) and (8) are exactly as shown in equation (10). In Columns (1), (5) and (7) we omit the interaction of temperature with the "poor country" dummy to obtain the average effect of temperature on emigration, averaging all countries. In Specifications (3) and (4) we also include a dummy called "prevalently agricultural" to denote countries in the top quartile of the distribution of agricultural value added as a share of GDP. This dummy and its interaction with the logarithm of temperature is used in place of Column (3) or together with Column (4), the interaction of temperature with the "poor country" dummy. Agricultural prevalence should be an alternative to GDP per person to identify poor countries, and to single out those on which temperature may have a strong impact on productivity via its effect on agriculture. This is an important check, as we presume agricultural productivity is the channel through which temperature affects migration. The number of observations varies between 114 and 116 countries over four decades, except when we include an interaction using the share of value added in agriculture (Columns 3 and 4), which reduces the number of observations significantly.

Two results emerge from Table 2. These results are consistent and robust across different specifications. The first is, when not including the interaction with the "poor country" dummy, Column (1) displays a non-significant effect of the average temperature on emigration rates for the full sample of poor and middle-income countries. Similarly, no significant effect is found on the precipitation variable. The second result, however, is that when we allow the coefficient on the temperature variable to vary between middle-income and poor countries (as we do in Column 2 and beyond) by adding an interaction with the "Poor country" dummy, the coefficient on temperature in middle-income countries (γ) turns positive and statistically significant at the 5% confidence level, while the coefficient of the interaction between the "poor country" dummy and the temperature (γ_P) becomes negative, quite large in absolute value, and significant at the 1% level. The net effect of temperature on emigration in poor countries, obtained by adding γ and γ_P , is reported in the second-to-last row of Table 2: it is also negative and statistically significant.⁸ The estimated coefficients in Column (2) indicate that a one percent increase in temperature increases international migration rates by four percent in middle-income countries, whereas it decreases emigration rates in poor countries by 16 percent, *ceteris paribus*. This implies a middle-income country with an average yearly temperature of 22 degree Celsius (the average of our sample) would experience a 20% increase in the rate of emigration if its average yearly temperature increased by one degree (roughly a 5% increase). Hence, at the average, this will imply an increase of the emigration rate from 0.042 to 0.05, with a 0.8 percentage point higher emigration rate. The same one degree Celsius warming in a poor country, however, would generate an 80% decrease in the rate of emigration (from 0.018 to 0.004). This seems a significant but reasonable impact. The only previous study that allows a comparison of magnitude for this effect is Cai et al (2014). In that study the basic specification (in their Table 2 Column 2) finds that an increase in temperature equal to one degree centigrade produces an increase in emigration rates to the average destination (and hence overall) by about 0.047 log points (i.e. 4.7%). This is an elasticity of the effect within one year. Our ten year elasticity for

⁸The "poor country" dummy identifies countries in the bottom of the country-of-origin income per capita distribution. This includes countries with income per person below \$1,500 in 1990 as "poor". In a robustness check (not reported), we use a less stringent definition of "poor" by including countries in the lowest tercile of the income distribution. This includes all countries with GDP per person below 2,000\$ as poor. The results, available upon request, are very similar to those reported in Table 2.

middle income countries is four time larger (20%), while for low income countries we obtain a negative elasticity. As emphasized above, Cai et al (2014) use gross rates and do not differentiate a response between poor countries and middle income ones, although the countries with large agricultural shares are likely relatively poor.

The coefficients on the variable "Precipitation" (δ) and Precipitation interacted with the "poor country" dummy (δ_P) are not statistically significant; we do not detect a comparable effect of precipitation on migration. Several other studies find small or non-significant effect of rainfall or flooding on the probability of migrating (e.g. Aufhammer and Vincent, 2012; Bohra-Mishra et al 2014; Mueller et al, 2014). We inquire further into this relationship by including only the precipitation variable in the regression, as warming can be related to increased probability of draught and act as a confounding factor. The estimates, reported in Table A1 of the appendix, show no significant correlation between precipitation and migration even when the variable "temperature" is omitted. Using specifications similar to those of Table 2, in fact, we observe that the estimated coefficient on the precipitation variable is never significant. According to these results, agriculture-related outmigration is mainly due to changes in temperature, rather than changes in precipitations.

If the negative effect on migration in poor countries proceeds from lower agricultural productivity and liquidity constraints, as assumed by our simple model, then it should be particularly strong for countries heavily depending on agriculture. Granted that there is a strong negative correlation between the share of agriculture in GDP and income per person, so that poor countries have, in general, a larger share of agriculture value added in GDP, we explicitly include a dummy in Column (3) capturing those countries with a large agricultural sector. Their productivity and incomes are likely to be more affected by warming temperatures. We compute a dummy for a country being "prevalently agricultural", which is equal to one if a country belongs to the top quartile in the world distribution of agriculture as a share of GDP.⁹ Columns (3) and (4) add interactions between temperature/precipitation and the agricultural dummy to Specification (10). The coefficients of the temperature-agricultural interaction are negative and statistically significant when included instead of the interactions with "poor" (Column 3) and even when included in addition to those variables (in Column 4). In particular, conditional on a country being poor, an increase in average temperature by 1% (about 0.2 degree Celsius at the sample average) decreases the rate of emigration by an additional 12 percent if the country is also highly agricultural-dependent. When included together, the "poor country" and "prevalently agricultural" dummies interacted with the temperature have similar coefficients. Finally, notice that different definitions of our

 $^{^{9}}$ As in the case of GDP per capita, the choice of the year for drawing the distribution was determined by the availability of data. For the agricultural share the year 2000 was chosen.

sample (non-OECD versus countries below the top quintile of GDP per person) and a different weighting of the temperature data do not make much of a difference in the estimates. Hence, we will mostly use the non-OECD definition of poor and middle-income countries and population weights.

In Table 3 we present some robustness checks that confirm the results in Table 2. In this case, we divide the sample and analyze the effects of temperature and precipitation on emigration for middle-income countries (Columns 1-4) and poor countries (Columns 5-6) separately. This implies that we allow all the coefficients in our model (not only those on average temperature) to differ between these two groups of countries. The point estimates in Table 3 are in line with the corresponding coefficients in Table 2. An increase in temperature by 1% in middle-income countries increases emigration rates by about 4%. The same average temperature increase in poor countries decreases emigration rates by 20%. The precipitation variable does not appear to have any significant effect on either group.

Another simple implication that the effects we are estimating proceed through the impact of climate on agricultural income and not from other omitted channels is that average temperature should not have any impact on the emigration rates from rich countries. Agriculture is not an important source of income for those countries and rural population in those countries is a small percentage of the total. Hence one would not expect either positive or negative migration effects of temperature in these countries. In general we do not inlcude rich countries of origin in our analysis, as agricultural productivity is much less relevant there (and possibly less subject to weather fluctuations). In column (7) of Table 2 we check that the irrelevance of temperature for net emigration rates in rich countries is confirmed by the data. In a specification similar to (1), including only OECD countries as origin, we find an estimate of the impact of temperature on emigration rates that is very small in value (1.04) and not significant at all.¹⁰

In Table 3 we also investigate whether the effect of temperatures on migration varies depending on the country being in the group of warmer (hotter than the median) or cooler (colder than the median) countries in the world within each income group. As the effect of our regressions (because of country fixed effects) are identified on the change in temperature over the considered decades, one might expect that countries starting with high temperatures might suffer worse consequences from average warming. Even if adaptation to increasing temperature is possible at any level of temperature by choosing the optimal crop mix and optimal mix of crops and animal activities, at high temperatures the overall profitability of agriculture declines (Mendelsohn et al., 1994; IPCC, 2014). For example, higher

 $^{^{10}}$ A similar non-significant effect is found when the sample of rich countries is selected on the basis of GDP per capita.

temperatures in the tropics reduce the size of the agro-climatic zones suitable for perennial crops. They also reduce the growing season length, influence the composition of farming systems towards livestock-dominated food production (IPCC, 2014), and force farmers to adopt climate-resistant crops. All these adaptation measures, however, prove to be less productive and therefore, at higher temperatures, average warming may be more detrimental. It may also be, however, that changes in temperature are damaging to any poor country's agricultural productivity, at least for a while, because local crops and production have been adapted and perfected for centuries to a certain yearly cycle of temperatures and a systematic alteration of that is damaging to productivity at any level of temperature. Moreover keep in mind that we are considering poor or middle income countries in which technological and scientific adjustment of agricultural practices to climate changes may be slow. Specifications (2), (4) and (6) include in the regression an interaction of the temperature variable with the dummy for countries that are warmer than the median. The stratification of the sample implies that (2) and (4) estimate the effect only on middle-income countries and (6) only on poor ones. In no case do we find significant interaction effects. Increases in temperature are equally damaging for agriculture both in cold and warm countries and they translate in more migration in middle-income countries while poor countries respond with less migration, possibly because of the tight liquidity constraint. This result is somewhat in contrast with Borha-Mishra et al (2014) who find a positive effect of temperature on emigration rates in Indonesian villages only for values above the median of 25 degree Celsius. Their identification is however based on variation over time, for one country only and their focus is on net emigration from a village rather than international emigration.

The opposite effects of temperature on the emigration rate in poor and middle-income countries is consistent with the model presented in section 3 and with simple economic logic that emphasize both the presence of incentives and constraints. If secular heating damages agricultural productivity, countries with a large dependence on agriculture and very low income experience a substantial worsening of their liquidity and hence of their ability to pay for emigration. On the other hand, middle-income countries experience a worsening of potential earnings but, as long as people can afford emigration costs, this increases their willingness to emigrate. The finding is clearly related to the widespread regularity (summarized for instance in Clemens, 2014) that emigration rates have a humpshaped relation with income per person in the country-of-origin. An increase in income in very poor countries allows them to pay the costs and increases emigration rates. Past a certain level, however, higher income reduces the incentives to migrate. Most studies find the inversion in this relation takes place between 3,000 and 5,000 \$ per person, which is a level in the low-range of the middle-income countries of our sample. Hence, consistently with

that literature, the negative effect we find on poor countries can be fully due to worsening of income and of liquidity constraints.

In Tables A2 and A3 of the Appendix we explore two further robustness checks. In Table A2 we consider decade differences in emigration rates and temperature and omit the country fixed effects in the panel. The estimated effects of temperature and its interaction with the "poor" dummy are somewhat attenuated relative to the estimates in levels, as can happen if differencing increases the noise-to-signal ratio. However, the positive impact on middleincome countries is still significant and the estimated coefficient on "poor countries" is still negative (although not significant). Table A3, instead, shows the results of a "long-difference" estimation. In this case we take differences (between temperature and emigration rates) over thirty years – between the seventies and nineties – and we estimate a cross-sectional regression in long differences. Comparing the long-run estimates of Table A3 and the corresponding medium-run coefficients of Table 2 we see that elasticities are larger in the second case. One may conclude that protracted episodes of temperature increase have stronger effects on the long-run propensity to migrate, especially in poor countries. This may imply that, by worsening income perspectives in rural areas, an increase in average temperatures may have damaging effects on income that amplify with time as the possibility of emigration becomes ever more remote. Overall, the pattern emerging consistently across specifications is that increased temperature encourages emigration in middle-income countries, but reduces it in poor countries. This effect is significant and for poor countries it may imply, especially in the long-run, a reduced ability of people to emigrate and escape poverty.

5.2 Effects on Urbanization

International migration is certainly a way to take advantage of economic opportunities and is also a way to escape local rural poverty. However, most of the population does not migrate internationally because of high costs, lack of information and limited opportunities (e.g. Pritchett, 2006). Internal migration, especially from rural areas to urban areas and cities, is an alternative. While the economic returns to internal migration are lower, it is less costly than emigration. The same ideas, developed in the model of section 3, can be applied to rural-urban migrants. Increased temperatures will affect agriculture productivity more than urban activities and, thus, will mainly affect the income of rural populations. Moreover, the returns to skills are likely to be larger in the city than in the countryside (as the model postulates). In very poor countries, the rural population may be so poor that it lacks the income to overcome the information and cost barriers for migrating to the city. This may actually be the main reason preventing migration in poor countries. Hence, a decrease in current income would make the transition to cities even less likely. In middleincome countries, instead, liquidity constraints may be less severe for rural workers, and so a worsening of their income perspectives in agriculture may increase inflow into cities. Thus, we can use a measure of "urbanization" as the outcome of interest, namely the share of total population of a country living in cities, whose change is mainly due to rural-urban migration. We analyze the effects of increasing average temperatures and precipitation on urbanization, just as we did for international migration rates.

The structure of Table 4 follows exactly that of Table 2, except that the outcome variable is the share of population living in urban areas relative to total population in the country. Columns (1)-(4) and (7)-(8) calculate the average temperature in a country by using population weights, while Columns (5)-(6) use area weights. Columns (1)-(6) use the non-OECD sample, while (7)-(8) use the countries in the bottom and fourth quintiles of GDP per person as the sample. Considering the results in Columns (1) and (2), the findings suggest a similar behavior of rural-urban migration as was found for international migration. Namely, an increase in average temperatures increases the degree of urbanization, speeding up the rural-urban transition in middle-income countries, but slows it down in poor countries. A 5%increase in temperature at the average (22 degree Celsius), equal to about one degree Celsius, increases urbanization rates by 4 percentage points in middle-income countries, while it decreases by the same in poor countries. Considering poor countries have an average urbanization rate of around 19% in this period, and the increase in urbanization was about 20 percentage points per decade over the considered period, an increase in temperature by one degree Celsius may slow the urbanization process very significantly. As in the case of international migration, precipitation does not appear to have a significant effect on urbanization. On the contrary, the interaction between precipitation and the "poor country" dummy has a marginally significant effect in three specifications. This effect, however, is not robust and only significant at the 10% level, so we do not think it is evidence of an additional productivity effect. In Column (3) we confirm the negative effect of temperature on rural-urban migration exists on "prevalently agricultural" countries (as much as on "poor" countries), but Column (4) shows that when we include both interactions, only the one with "poor" remains significant.

Some studies (e.g. Marchiori et al, 2012) postulate a direct connection between ruralurban migration and international migration. Namely, they consider rural populations affected by weather shocks as first moving into cities, increasing urbanization rates and crowding urban centers. International migration proceeds from this as a consequence of crowding and a decrease in income/amenities in cities. This is a reasonable possibility and our results are consistent with it. However, rural populations may directly migrate abroad, and the crowding of cities may decrease income (crowding), or increase it (agglomeration), so that international migration may not be a direct consequence of urbanization. In our framework, we consider urbanization and international migrations as two possible outcomes, both driven by a decline in rural income, but not necessarily sequential or linked. Individuals in rural areas are more affected by temperature and, hence, their migration behavior to cities or abroad could be affected. The results from Table 4 show increases in temperature have the same type of effects on rural-urban migration as on international emigration, strengthening the plausibility of our interpretation.

5.3 Where do People Migrate in Response to Warming?

Are rich and far away countries the main destination for people who move out of middleincome countries as a consequence of warming and lower agricultural productivity? Or are these individuals prevalently moving to nearby countries that are experiencing somewhat better economic opportunities? Does warming produce a large scale movement of individuals from middle-income countries in Africa, Asia and Latin America to rich countries in Europe and North America? Or does it produce a more local reallocation across middleincome countries within a region? In order to analyze this question, we compute the net emigration rates for each country of our sample, separating emigrants to OECD and non-OECD countries. Table 5 presents the findings using this split. We then separate emigration between destinations within 1,000 Km from the origin and farther destinations and we calculate separate rates for short-distance and long-distance emigration. Table 6 shows the estimates in this case.

The estimates in Table 5 show a very interesting pattern of temperature-induced emigration. Columns 1-4 show the effect on emigration rates to non-OECD destinations. Columns 5-8 show the impact on emigration rates to OECD destinations. Inclusion of the interaction between temperature and the Poor-country dummy varies by column. Two findings are worth noting. First, increasing in temperatures increases emigration rates from middleincome countries only to non-OECD destinations. It has no significant effect on emigration to OECD countries. This result is consistent with our proposed channel, as emigration is driven by a worsening of local opportunities and not by an increase in opportunities in rich countries. Hence, migrants move to where they have better chances of finding a job given their current constraints. This "push" factor (decreased rural income) increases migration to similar economies rather than to OECD economies. On the other hand, the immigrationreducing effect for poor countries (due to worsening opportunities) affects both types of destinations, as potential emigrants become less likely to leave the country. Similarly and consistently with those estimates, Table 6 – whose structure mirrors that of Table 5 – shows higher temperatures increase emigration rates to close destinations (<1000 Km.) for middleincome countries, while poor countries experience a decrease of emigration rates to any other country (the standard error on the estimated effects in Columns 2 and 4 are rather large, however). Combining the effect on poor and middle-income countries, it appears that increases in average temperatures may actually decrease overall emigration to OECD countries. Middle-income countries are not more likely to experience emigration towards those destinations, while poor countries experience a reduction in emigration rates. This is bad news for the potential income of individuals from non-OECD countries, for which emigration to rich countries constitutes one of the best options for increasing their household income and economic well-being.¹¹

6 Robustness Checks

We conducted a variety of robustness checks and present the results in the Appendix at the end of the paper. However, we will summarize them here. First, we have focused our attention on countries in sub-Saharan Africa (SSA). This specific group includes most of the poor countries and also those whose productivity is more likely to be affected by temperature increases. Moreover, several of the previous studies (e.g. Barrios et al 2012 and Marchiori et al 2006) have considered only sub-Saharan African Countries in their analysis of extreme weather events and migration. In Column (1) of Table A4 we have included a dummy, "SSA", in the place of "poor" and an interaction term between temperature (and precipitation) and the "SSA" dummy. In Column (2) we have included both the SSA and the "poor country" dummy, as well as their interactions. Columns (1) and (2) analyze the effect on emigration rates, while Columns (3) and (4) look at the impact on urbanization rates. A variety of interesting results emerge, mostly confirming the previous ones. First, temperature has a negative impact on emigration out of SSA countries and is mainly due to them being poor and primarily rural countries. In Column (1) the interaction is negative and significant, while in Column (2) we find no difference for sub-Saharan Africa once we allow for a different coefficient on poor countries. Hence, our analysis shows that SSA countries' migration response to temperature increases looks in line with the response of other poor countries: hotter climate reduces the ability of rural populations to migrate. A second effect, however, is now estimated as significant and is consistent with the temperature effect:

¹¹In an alternative (unreported) way of testing the effect of warming on the destination distance, we have calculated the "average distance" of emigrants, where the weighted distance variable is given by: $\sum_{j} dist_{ji} * \frac{flows_{ji}}{\sum_{j} flows_{ji}}$, where $dist_{ji}$ is the geographical distance between origin j and destination i. Increases in temperature significantly decreases the average distance of emigration from middle-income countries.

a decrease in Ln(precipitation). In other words, drier climates are associated with higher emigration rates for non-SSA countries and lower emigration-rates for SSA countries. These effects are significant at the 10% level and are robust to the inclusion of the "poor" dummy interactions, revealing a role for precipitation as a driver of agricultural productivity in SSA. This may be due to the fact that SSA countries are affected by drier climate because of their rural nature, or that investments in irrigation are rare among farm households in this area of the world. Moreover, this finding indicates that some regions of frequent droughts (e.g., East Africa) are extremely dependent on seasonal water. A decline in precipitation and an increase in temperature (both of which would be associated with lower agricultural productivity in this region) will drive lower emigration, leaving people in a worsening condition of poverty. Once we allow for this effect in SSA countries, higher temperature and lower precipitation push emigration from the other countries that are not as poor. So, in line with their extreme poverty and dependence on agriculture, as well as consistent with the fact that emigration rates out of SSA countries are rather low, worsening agricultural productivity coming from warmer and drier weather reduces rural-urban migration and emigration from sub-Saharan Africa.

Table A5 includes a variable that captures the realization of extreme temperatures during a decade in a country. The variable measures the number of years of a decade in which the temperature was above or below two standard deviations of the 1960-2000 period mean for the country. Temperature anomalies, both above and below the mean, have been more likely in the most recent decade, namely between 1990 and 2000. The occurrence of one extreme episode in the decade was registered 117 times. Out of these, 58 cases happened between 1990 and 2000, 21 between 1990 and 1980, 23 between 1970 and 1980 and 15 between 1960 and 1970. For some countries, these temperature anomalies have occurred twice and three times within a single decade, but these occurrences were rather rare.

We added to the basic Specification the count of the extreme temperature events per decade as well as its interaction with the "poor country" dummy variable. Columns (1)-(6) of Table A5 show that emigration rates in middle- income countries are not influenced by the occurrence of temperature anomalies, nor do they matter for migration in poor countries, regardless of the specification of countries we use. Columns (7) and (8) focus specifically on extremely high temperature years (only anomalies in the high temperature range), but again we do not find significant effects.

Given that climate change is expected to bring an intensification of natural disasters, such as droughts, floods, storms and extreme heat, we count the total number of natural disasters in each specific decade, computed from the international Disaster Database (Guha-Sapir et al., 2015). In Table A6 we added this variable along with its interaction with the

"poor country" dummy variable. We find that emigration rates are not influenced by the occurrence of natural disasters. In further specifications (not reported but available upon request) we have included each type of natural disatser, namely droughts, floods, storms and extreme heat, individually in the regression as count of their occurrences in the decade. We do not find, however, any impact. It is possible that natural disasters and long-term changes in temperature and rainfall drive different types of migration, more akin to local mobility and potentially reversed in years of good weather. Hence natural disasters may be responsible for the displacement of people in near areas, generating nonpermanent transitions, but in the long run, as they are rare and they only occurr in some countries, they may not affect significantly rural-urban and international migration when looking at all countries. This finding is consistent with the analysis of Beine and Parsons (2015) who do not find direct impact of the same type of events on bilateral migration and with the analysis of Bohra-Mishra et al (2014) for the migration behaviour in Indonesian villages. This last paper finds a permanent migration response in Indonesia (a middle income country) to long-run increases in temperature but not to episodic disasters. Mueller et al. (2014) report that heat stress but not flooding has a significant effect on migration in Pakistan.

Finally, Table A7 specifies temperature in levels, rather than logarithms. The estimated values are comparable to those obtained using the log specification. An increase in one degree Celsius would increase emigration rates by 27% in middle-income countries (using Column (2) coefficients) and decrease emigration rates by 86% in poor countries. This is similar to the results reported in Table 2.

7 Effects on Structural Change and GDP

In the previous sections we have estimated a reduced-form relationship between temperature (and precipitation) and emigration/urbanization rates across countries. We have shown these correlations are consistent with the following interpretation: increased temperatures decrease agricultural productivity and exacerbates the liquidity constraint for rural populations in poor countries, reducing their ability to emigrate, but increases incentives for rural populations in middle-income countries to emigrate. While several checks confirmed this interpretation is plausible, and while the "exogenous" nature of temperature and its variation across countries relative to local economic and social conditions ensure that reverse causality problems are likely limited, it is hard to really identify the exact channels of the estimated effects. One could argue higher temperatures have other disruptive effects in poor countries besides their impact on agriculture (increased conflict, wars, affects on health and fertility) that also reduce emigration rates. Admitting it is hard to identify those channels fully and precisely, we want to emphasize the agricultural productivity channel. Hence, we will test several other plausible implications derived from our model, in which the migration response (or lack of it) is prevalently linked to rural income (agricultural productivity). Higher temperatures influence the emigration rates by lowering agricultural productivity both in poor and in middle-income countries. While in middle-income countries lower agricultural productivity translates into higher emigration rates, and hence a further reduction in agricultural value added, the existence of liquidity constraints in poor countries implies that lower agricultural productivity prevents people from leaving and, hence, this second channel of potential decline in agriculture value added as share of GDP is muted. This would imply that higher temperatures should have a negative impact on value added in agriculture as a share of GDP, and that this effect should be particularly strong in middle- income countries in which emigration also causes a decline in the number of agricultural workers. To test this hypothesis, we regress a reduced-form relationship between temperature and agricultural value added as a percent of GDP and show the coefficient estimates in Table 7. As before, we consider only middle-income and poor countries and obtain the data on value added in agriculture as a share of GDP from the World Development Indicators (World Bank, 2015). In line with expectations, increases in temperature significantly decrease the agricultural share of GDP for middle-income countries (Columns (2) and (4)). However, for poor countries the effect is more imprecisely estimated and not significantly different from 0 (although positive in point estimate, see last row of the Table). This is consistent with the idea that in middle-income countries the direct effect of warming (producing a decrease in agricultural productivity) and the indirect effect (inducing migration of rural population to cities or abroad) both contribute to reduced value added in agriculture. In poor countries, to the contrary, only the direct effect is present and in aggregate may be less significant. Therefore, lower agricultural productivity because of higher temperatures, combined with the possibility of migrating (to cities or abroad) may simply speed up the structural transformation of some middle-income countries away from rural economies toward more urban and productive economies. Though, in very poor countries where migration mechanisms do not work, the loss in agricultural productivity does not trigger a structural change from rural to urban economies.

The previous channel, operating through a structural transformation, encouraged or slowed by warming, suggests another implication of our theory. Warming could be associated with an increase in GDP per capita in middle- income countries where rural workers move to more productive cities. Instead, it should be associated with a decrease in poor countries where rural workers are stuck in an impoverished agricultural sector. We test this implication in Table 8 in which the dependent variable is the logarithm of GDP per capita, obtained from the Penn World Table (2009) in Columns (1)-(2) and (5)-(6) or from the World Development Indicators (World Bank, 2015) in columns (3)-(4) and (7)-(8). Consistently with these predictions, middle-income countries experience growth in GDP per person (Columns (2), (4), (6), and (8)). The mobility of workers into cities (with higher productivity potential) and the out-migration of poor rural workers result in a positive effect, significant at the 5% level, of between 1.4 and 2.3. To the contrary, poor countries experience a negative impact of higher temperatures on income per person, with an interaction term that is negative and significant.

Overall, warming temperatures increase GDP per capita in middle-income countries while negatively affecting GDP per capita in poor countries. While a similar result was also found by Dell et al 2012, and hence it is not new, our model and previous analysis provides an additional explanation, linked crucially to the role of mobility/migration as a margin of adjustment to these weather/agricultural productivity shocks. In countries in which agricultural productivity is not so low as to be at subsistence level, a worsening of economic opportunities in agriculture pushes individuals to migrate to cities and to other countries, opening them up to better opportunities and eventually helping to raise the average income of a country. Urbanization and moving out of agriculture are crucial mechanisms to the increase of GDP, and in countries at middle-income levels, warming can be an additional push to realize these gains. However, in places in which agricultural productivity is so low as to leave rural populations liquidity constrained and limited to agriculture, then warming and subsequently lower agricultural productivity may slow economic transformation and growth. These effects ultimately contribute to a poverty trap.

8 Discussion and Conclusions

In this paper we have focused on the potential impact the increase in average temperature, experienced in many countries during the last few decades, may have on internal and international migration. We have assumed the main impact of temperature increase is through an effect on agricultural productivity and, hence, countries experiencing larger increases may have suffered decline in agricultural productivity. This channel, which should mainly affect rural populations, has a varied consequence on emigration rates depending on the income level of potential migrants. In very poor countries, where the main obstacle to migration is that people are so poor they cannot afford the cost of emigration, warming and lower rural income may imply less emigration. That is, rural populations go even deeper into poverty and subsistence mode as a consequence of low agricultural productivity. In countries where income is not as low, however, lower agricultural productivity will enhance the incentives to migrate either to cities or abroad. Consistent with these predictions, we find climatic warming is associated with significantly higher emigration rates in middle-income countries and significantly lower rates in poor countries (income per capita lower than 1,500 \$ per person, which includes many sub-Saharan Africa countries). We also show, as a consequence of the migration out of rural poverty encouraged by warming, middle-income countries are better off in terms of their GDP per capita. Poor countries, on the contrary, are made worse off and may be further trapped in poverty as a consequence of climatic warming.

References

Auffhammer M. and J. R. Vincent (2012) "Unobserved time effects confound the identification of climate change impacts" *Proceedings of the National Academy of Sciences.* vol. 109(30), pages 11973-11974.

Auffhammer M., S. M. Hsiang, W. Schlenker and A. Sobelz (2013) "Using Weather Data and Climate Model Output in Economic Analyses of Climate Change" Review of Environmental Economics and Policy, 7(2), 181-198

Barrios Salvador, Luisito Bertinelli and Eric Strobl (2006) "Climatic change and rural-urban migration: The case of sub-Saharan Africa" *Journal of Urban Economics*, vol. 60(3), pages 357-371.

Beine M. and C. Parsons (2015) "Climatic Factors as Determinants of International Migration" The Scandinavian Journal of Economics, 117(2): 723–767

Bohra-Mishra Pratikshya, Michael Oppenheimer and Solomon M. Hsiang (2014) "Nonlinear permanent migration response to climatic variations but minimal response to disasters" *Proceedings of the National Academy of Sciences*, vol 111(27), pages 9780-9785.

Borjas G. J. (1987) "Self-Selection and the Earnings of Immigrants", *The American Economic Review*. 77(4): 531-553

Cai, Ruohong, Shuaizhang Feng, Mariola Pytliková and Michael Oppenheimer, (2014) "Climate Variability and International Migration: The Importance of the Agricultural Linkage," CReAM Discussion Paper Series 1418, Centre for Research and Analysis of Migration (CReAM), Department of Economics, University College London.

Clemens M. (2014) "Does Development reduce Migration?" IZA Discussion Papers 8592, Institute for the Study of Labor (IZA)

CIESIN - Columbia University, United Nations Food and Agriculture Programme - FAO, and Centro Internacional de Agricultura Tropical - CIAT (2004) Gridded Population of the World, Version 3 (GPWv3): Population Count Grid. Palisades, NY: NASA Socioeconomic Data and Applications Center (SEDAC)

Cline, William R., 2007, "Global Warming and Agriculture: Impact Estimates by Country", Washington: Center for Global Development and Peterson Institute for International Economics.

Dell M., B. Jones and B. Olkien (2012) "Temperature Shocks and Economic Growth: evidence from the last half Century" *American Economic Journal: Macroeconomics*, 4(3): 66–95.

Dell M., B. Jones and B. Olkien (2014) "What Do We Learn from the Weather? The New Climate-Economy Literature," *Journal of Economic Literature*, 52(3): 740-98

Dillon Andrew; Valerie Mueller and Sheu Salau (2011) "Migratory Responses to Agricultural Risk in Northern Nigeria" *American Journal of Agricultural Economics*, vol. 93(4), pages 1048-1061

Docquier, F. & Marfouk, A. & Özden, C. & Parsons, C. (2011) "Geographic, Gender and Skill Structure of International Migration," MPRA Paper 47917, University Library of Munich, Germany.

Feng Shuaizhang, Alan B. Krueger and Michael Oppenheimer (2010)"Linkages among climate change, crop yields and Mexico-US cross-border migration" *Proceedings of the National Academy of Sciences*, vol. 107(32), pages 14257-14262.

Gray C. L. (2009) "Environment, Land, and Rural Out-migration in the Southern Ecuadorian Andes" World Development, 37(2): 457-468

Gray C. L. and Richard Bilsborrow. (2013) "Environmental influences on human migration in rural Ecuador" *Demography* 50(4): 1217-1241.

Gray C. L. and Valerie Mueller (2012a) "Drought and population mobility in rural Ethiopia" World Development 40(1): 134–145

Gray C. L. and Valerie Mueller (2012b) "Natural disasters and population mobility in Bangladesh" Proceedings of the National Academy of Sciences, 109(16)

Grogger, Jeffrey, and Hanson, Gordon H. (2011) 'Income maximization and the selection and sorting of international migrants', *Journal of Development Economics*, 95/1, 42-57.

Guha-Sapir D., R. Below, Ph. Hoyois - EM-DAT (2015) International Disaster Database – www.emdat.be – Université Catholique de Louvain – Brussels – Belgium

Hatton, T.J. & J.G. Williamson (1994). What Drove the Mass Migrations from Europe in the Late Nineteenth Century? *Population and Development Review* 20 (3): 533–559

Hatton, T.J. and J.G. Williamson (2003) "Demographic and Economic Pressure of Emigration out of Africa" *The Scandinavian Journal of Economics* Volume 105, Issue 3, pages 465–486, September 2003

Hatton, T.J. & J.G. Williamson (2011). "Are third world emigration forces abating?" World Development 39(1): 20-32.

Henry S., B. Schoumaker and C. Beauchemin (2004) "The impact of rainfall on the first out-migration: A multilevel event-history analysis in Burkina Faso" Population and Environment, vol. 25(5), pages 423-460.

IPCC (2014) "Fifth Assessment Report: Impacts, Adaptation, and Vulnerability" UNEP, Washington DC.

Jones B. and B. A. Olken (2010) "Climate Shocks and Exports," *American Economic Review*, vol. 100(2), pages 454-59

Kelley Co. P., Shahrzad M., M. A. Cane, R. Seager and Y Kushnir (2015) "Climate change in the Fertile Crescent and implications of the recent Syrian drought", Proceedings of the National Academy of Sciences, vol. 112(11), pages 3241-3246.

Marchiori L., Maystadt J.F and I. Schumacher (2012) "The impact of weather anomalies on Migration in Sub-Saharan Africa" *Journal of Environmantal Economics and Management*, 63: 355-374.

Matsuura, Κ, and С. J. Willmott (2007)"Terrestrial Air Temperature and Precipitation: 1900 - 2006Gridded Monthly Time Series, Version 1.01"http://climate.geog.udel.edu/~climate/

Mayda A.M. (2010) "International migration: A panel data analysis of the determinants of bilateral flow" *Journal of Population Economics*, October 2010, 23 (4): pp. 1249-1274.

Mendelsohn R.; W. D. Nordhaus and D. Shaw (1994) "The Impact of Global Warming on Agriculture: A Ricardian Analysis" *American Economic Review* 84(4): 753-771.

Mueller Valerie A., Clark L.Gray and Katrina Kosec (2014) "Heat stress increases long-term human migration in rural Pakistan", in Nature Climate Change, vol. 4, farm types and irrigation as an adaptation to climate change 182-185.

Munshi Kaivan (2003) "Networks in the modern economy. Mexican migrants in the US labor market" *Quarterly Journal of Economics*, vol. 118(2), pages 549-599.

Ortega F. and G. Peri (2012) "The Effect of Income and Immigration Policies on International Migration." *Migration Studies*, Volume 1, page 1-28; Oxford University Press. DOI: 10.1093/migration/mns004.

Ozden C.; C. R. Parsons; M. Schiff and T. L. Walmsley (2011) Where on Earth is Everybody? The Evolution of Global Bilateral Migration 1960–2000. World Bank Economic Review, 25(1), pages: 12 - 56

Penn World Table (2009) Version 6.3, http://datacentre.chass.utoronto.ca/pwt/

Pritchett Lant (2006) "Let their people come: breaking the gridlock on international labor mobility" Center for Global Development, Washington DC.

Roy, A.D. (1951) "Some thoughts on the distribution of earnings" Oxford Economic Papers, 3, (2) : 135-46.

UN (2014) "World Urbanization Prospects: The 2014 Revision", Department of Economic and Social Affairs, Population Division, United Nations publication

World Bank (2015) "World Development Indicators" http://data.worldbank.org/data-catalog/world-development-indicators

Zelinsky W. (1971) "The Hypothesis of the Mobility Transition." *Geographical Review* 61 (2): 219-249.

A Data Appendix

The migration data used in this paper are taken from Ozden et al. (2011) and include matrices of bilateral migrant stocks in five available census years spanning 1960–2000, for 226 countries of origin and 226 countries of destination. To compute the emigration rates used in the estimations, first we compute bilateral emigration net flows as differences between bilateral stocks in two consecutive censuses. Then we sum all bilateral flows for the same countries of origin j, setting negative values to 0, as they are likely be due to mortality of the stock of emigrants abroad. The emigration rate from country of origin j is the ratio between the aggregated net flows from origin country j and the origin country population at the beginning of the decade. The computed emigration rates span the period from 1970 to 2000.

The temperature and precipitation data are taken from Dell et al. (2012). The (terrestrial) monthly mean temperature and precipitation data at 0.5X0.5 degree resolution obtained from weather stations (Matsuura and Willmott, 2007) are aggregated into countryyear averages using the population in 1990 at 30 arc second resolution as weights or alternatively using area weights.

By merging the two datasets and considering only "Poor and Middle-income" countries, we were able to compile final datasets with 114, 115 and 116 countries. The exact number depends on the weights used to aggregate the weather station data (population or area), and on the way we defined "Poor or Middle-income" countries, whether by excluding OECD countries, or by considering the country GDP per capita.

Given that the emigration rates were only available at decade level, temperature and precipitation have been averaged over the 10 years of the decade. For almost all countries the data were available for four decades. Only for Namibia, the first decade available is 1990.

List of poor countries

Afghanistan, Benin, Burkina Faso, Burundi, Cambodia, Central African Republic, Democratic Republic of Congo, the Equatorial Guinea, Ethiopia, Gambia, Ghana, Guinea-Bissau, Lao People's Democratic Republic, Lesotho, Liberia, Madagascar, Malawi, Mali, Mozambique, Nepal, Niger, Nigeria, Rwanda, Somalia, Sudan, United Republic of Tanzania, Togo, Uganda, Yemen and Zambia

List of middle-income countries (population weights and excluding OECD countries)

Albania, Algeria, Angola, Argentina, Bahamas, Bangladesh, Belize, Bhutan, Bolivia, Botswana, Brazil, Brunei Darussalam, Bulgaria, Cameroon, Cape Verde, Chad, China, Colombia, Comoros, Congo, Costa Rica, Cote d'Ivoire, Cuba, Cyprus, Djibouti, Dominican Republic, Ecuador, Egypt, El Salvador, Fiji, Gabon, Guatemala, Guinea, Guyana, Haiti, Honduras, India, Indonesia, Iran, Iraq, Jamaica, Jordan, Kenya, Kuwait, Lebanon, Libya, Malaysia, Mauritania, Mauritius, Morocco, Namibia, Nicaragua, Oman, Pakistan, Panama, Papua New Guinea, Paraguay, Peru, Philippines, Puerto Rico, Qatar, Romania, Russia, Saint Vincent and the Grenadines, Samoa, Sao Tome and Principe, Saudi Arabia, Senegal, Serbia and Montenegro, Sierra Leone, Solomon Islands, South Africa, Sri Lanka, Suriname, Swaziland, Syrian Arab Republic, Thailand, Trinidad and Tobago, Tunisia, United Arab Emirates, Uruguay, Vanuatu, Venezuela, Viet Nam, Zimbabwe

List of middle-income countries (population weights and excluding top income countries according to GDP pc)

Albania, Algeria, Angola, Argentina, Bangladesh, Belize, Bhutan, Bolivia, Botswana, Brazil, Bulgaria, Cameroon, Cape Verde, Chad, Chile, China, Colombia, Comoros, Congo, Costa Rica, Cote d'Ivoire, Cuba, Cyprus, Djibouti, Dominican Republic, Ecuador, Egypt, El Salvador, Fiji, Gabon, Greece, Guatemala, Guinea, Guyana, Haiti, Honduras, India, Hungary, Indonesia, Iran, Iraq, Jamaica, Israel, Jordan, Kenya, Republic of Korea, Lebanon, Libya, Malaysia, Mauritania, Mauritius, Morocco, Mexico, Namibia, Nicaragua, Pakistan, Panama, Papua New Guinea, Paraguay, Peru, Philippines, Poland, Portugal, Romania, Russia, Saint Vincent and the Grenadines, Samoa, Sao Tome and Principe, Senegal, Serbia and Montenegro, Sierra Leone, Solomon Islands, South Africa, Sri Lanka, Suriname, Swaziland, Syrian Arab Republic, Thailand, Trinidad and Tobago, Tunisia, Turkey, Uruguay, Vanuatu, Venezuela, Viet Nam, Zimbabwe.

List of middle-income countries (area weights and excluding OECD countries)

Albania, Algeria, Angola, Argentina, Bahamas, Bangladesh, Belize, Bhutan, Bolivia, Botswana, Brazil, Brunei Darussalam, Bulgaria, Cameroon, Cape Verde, Chad, China, Colombia, Comoros, Congo, Costa Rica, Cote d'Ivoire, Cuba, Cyprus, Djibouti, Dominican Republic, Ecuador, Egypt, El Salvador, Fiji, Gabon, Guatemala, Guinea, Guyana, Haiti, Honduras, India, Indonesia, Iran, Iraq, Jamaica, Jordan, Kenya, Kuwait, Lebanon, Libya, Malaysia, Mauritania, Mauritius, Morocco, Mongolia, Namibia, Nicaragua, Oman, Pakistan, Panama, Papua New Guinea, Paraguay, Peru, Philippines, Puerto Rico, Qatar, Romania, Samoa, Sao Tome and Principe, Saudi Arabia, Senegal, Serbia and Montenegro, Sierra Leone, Solomon Islands, South Africa, Sri Lanka, Suriname, Swaziland, Syrian Arab Republic, Thailand, Trinidad and Tobago, Tunisia, United Arab Emirates, Uruguay, Vanuatu, Venezuela, Viet Nam, Zimbabwe.

Tables and Figures

Countries Included in The Complex	Non-OECD San	nple		Non-OECD Sample		
	Middle-Income	e Countries	Poor Countries			
Variable	Observations	Mean	Std. Dev.	Observations	Mean	Std. Dev.
Emigration rate (emigration flows/population)	338	0.042	0.084	120	0.018	0.020
Temperature, °C (pop weight)	338	22.118	4.925	120	23.499	4.172
Precipitation, 100s mm/year (pop weight)	338	13.406	8.818	120	11.407	5.157
Temperature, °C (area weight)	330	22.334	5.037	120	23.606	4.200
Precipitation, 100s mm/year (area weight)	330	13.231	9.229	120	11.033	5.695
Share of Urban Population	420	0.422	0.222	145	0.194	0.112
Emigration rate (to non-OECD destinations)	338	0.014	0.034	120	0.014	0.018
Emigration rate (to OECD destinations)	338	0.028	0.073	120	0.004	0.004
Emigration rate (to close destinations)	289	0.009	0.037	104	0.010	0.018
Emigration rate (to distant destinations)	338	0.033	0.065	120	0.009	0.011
Agriculture, value added (% of GDP) (WDI source)	242	16.298	11.147	83	34.787	11.992
GDP per capita, constant, PPP (Penn World Table source)	332	8'197	12'896	114	1'167	776
GDP per capita, constant, local currency unit (WDI source)	290	467'717	1′512′382	96	179'070	419'531

Table 1: Summary Statistics

Note: The first three columns of the table show the summary statistics including as country of origin of immigrants non-OECD countries, excluding those in the bottom quartile of the GDP per capita distribution. The remaining three columns show the summary statistics for the sample of non-OECD countries in the bottom quartile of the per-capita GDP distribution. The sample is supposed to include countries of the world that are "Poor" or "Middle Income".

Table 2: Temperature and Emigration

Dependent Variable= Ln(Emigration/Population)	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
		Population we Countri	eights. Non-OE es of origin	CD	Area weigh Countrie	ts. Non-OECD es of origin	Populati Countri exclude qu	on weights. es of origin top income ıintile
Ln(Temperature)	1.931	3.755**	2.695	3.836**	0.597***	0.627***	2.689	4.398***
	(1.892)	(1.661)	(1.904)	(1.790)	(0.074)	(0.064)	(1.746)	(1.224)
Ln(Temperature) X Poor		-19.967***		-17.546***		-17.203***		-20.134***
		(6.607)		(5.068)		(6.369)		(7.118)
Ln(Temperature) X prevalently agricultural			-23.996***	-15.939*				
			(8.457)	(8.285)				
Ln(Precipitation)	-0.309	-0.223	-0.032	-0.113	0.057	-0.018	-0.369	-0.276
	(0.352)	(0.325)	(0.396)	(0.395)	(0.350)	(0.342)	(0.422)	(0.393)
Ln(Precipitation)X Poor		-1.399		-0.373		-0.543		-1.313
		(1.912)		(2.623)		(1.978)		(1.921)
Ln(Precipitation) X prevalently agricultural			-2.246	-1.674				
			(1.423)	(1.577)				
Country of origin Fixed Effects	yes	yes	yes	yes	yes	yes	yes	yes
Decade X Region effects	yes	yes	yes	yes	yes	yes	yes	yes
Decade X Poor effects	yes	yes	yes	yes	yes	yes	yes	yes
Observations	458	458	414	414	450	450	462	462
R-squared	0.179	0.201	0.202	0.216	0.186	0.204	0.195	0.218
Number of countries of origin	115	115	104	104	114	114	116	116
Temperature effect on poor countries		-16.212**		-13.711*		-16.576**		-15.736**
Temperature effect on agricultural countries			-21.301**	-12,103				

Poor and Middle-Income countries of origin included, years 1970-2000.

Note: Each column corresponds to a different Least Square estimated regression with fixed effects. The sample of countries for columns 1-6 are all non-OECD countries. In columns 1-4 the weather station data are averaged using population weights. Columns 5-6 use area as weight. Columns 7-8 use a sample of poor/middle-income countries of origin in the bottom to the fourth quintiles in the per-capita GDP distribution. The standard errors are cluster by country of origin. *, **, *** indicate significance at the 10, 5 and 1% confidence level.

Table 3: Temperature and Emigration

Separate estimation for Poor and Middle-Income countries of origin 1970-2000

Dependent Variable= In(Emigration rates).	(1)	(2)	(3)	(4)	(5)	(6)	(7)
	Non-OECD Middl	e Income Countries	Middle-income Co excluding top and quintile of GDP pe	untries, bottom r person	Poor countries: bc GDP per person in	ottom quartile of Non-OECD sample	Rich countries (OECD)
In Temperature	3.801**	3.933**	4.523***	4.179***	-21.531***	-17.661***	1.045
	(1.742)	(1.762)	(1.277)	(1.324)	(6.831)	(5.858)	(2.22)
In Temp. interacted with hot		-1.695		4.815		14.475	
		(5.336)		(4.814)		(14.972)	
In Precipitations	-0.253	-0.235	-0.306	-0.119	-1.617	1.595	-0.649
	(0.326)	(0.433)	(0.390)	(0.650)	(2.371)	(2.225)	(0.644)
In Prec. Interacted with wet		-0.041		-0.332		-5.676***	
		(0.645)		(0.756)		(1.965)	
Country of origin Fixed Effects	yes	yes	yes	yes	yes	yes	yes
Decade X Region effects	yes	yes	yes	yes	yes	yes	yes
Observations	338	338	342	342	120	120	120
R-squared	0.225	0.226	0.256	0.259	0.250	0.312	0.28
Number of countries of origin	85	85	86	86	30	30	30
Temp. effect in hot countries		2.238		8.994*		-3.186	

Note: Each column corresponds to a different Least Square estimated regression with fixed effects. The sample of countries for columns 1-2 are non-OECD countries, excluding those in the bottom quartile of GDP per capita distribution. Columns 3-4 include countries that are not in the top or bottom quintile of the world GDP per capita distribution. Columns 5-6 use countries of origin in the bottom quartile of the per-capita GDP distribution. Column (7) includes only OECD countries. The weather station data are averaged using population weights. The standard errors are cluster by country of origin. *, **, *** indicate significance at the 10, 5 and 1% confidence level.

Table 4: Temperature and Urbanization

Dependent Variable= Urban Population as share of total	(1)	(2)	(3)	(4)	(5)	(6)	(7)		(8)
	Ро	pulation weig Countries	ghts. Non-OE s of origin	CD	Area wei Count	ghts. Non-OECE ries of origin	Populatic origin excl	on weig ude to	hts. Countries of p income quintile
Ln(Temperature)	0.376	0.863*	0.656	0.908*	0.495	0.918*	0.165	'	0.455
	(0.342)	(0.461)	(0.422)	(0.496)	(0.398)	(0.482)	(0.274)		(0.356)
Ln(Temperature) X Poor		-1.661***		-1.365**		-1.785***			-1.175**
		(0.566)		(0.634)		(0.596)			(0.476)
Ln(Precipitation)	-0.017	0.003	-0.033	-0.033	-0.030	-0.020	0.001		0.024
	(0.037)	(0.039)	(0.043)	(0.044)	(0.036)	(0.038)	(0.039)		(0.044)
Ln(Precipitation)X Poor		-0.156*		-0.125		-0.169**			-0.159*
		(0.087)		(0.097)		(0.080)			(0.086)
Ln(Temperature) X prevalently agricultural			-1.580***	-0.824					
			(0.523)	(0.579)					
Ln(Precipitation) X prevalently agricultural			0.001	0.091					
			(0.073)	(0.083)					
Country of origin Fixed Effects	yes	yes	yes	У	ves	yes	yes	yes	yes
Decade X Region effects	Yes	Yes	Yes	Y	′es	Yes	Yes	Yes	Yes
Decade X Poor effects	yes	yes	yes	У	ves	yes	yes	yes	yes
Observations	565	565	510	510	448	448	570		570
R-squared	0.723	0.733	0.723	0.727	0.689	0.702	0.753		0.759
Number of countries of origin	114	114	103	103	113	113	115		115
Temp, effect on poor countries		-0.798**		-0.457		-0.867**			-0.720**

Poor and Middle-Income countries of origin included, years 1960-2000.

Note: Each column corresponds to a different Least Square estimated regression with fixed effects. The sample of countries for columns 1-6 are all non-OECD countries. In columns 1-4 the weather station data are averaged using population weights. Columns 5-6 use area as weight. Columns 7-8 use a sample of poor/middle-income countries of origin in the bottom to the fourth quintiles in the per-capita GDP distribution. The standard errors are cluster by country of origin. *, **, *** indicate significance at the 10, 5 and 1% confidence level.

Table 5: Temperature and Emigration, OECD versus non OECD destinations

	P	oor and Middle-Income	countries of origin included	. vears 1970-2000.
--	---	-----------------------	------------------------------	--------------------

Dependent Variable= In(Emigration rates).	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
		NON OECD	destinations		OECD destinations			
	Nor	Non-OECD Countries of origin		es of origin	Nor	I-OECD	Countries of origin exclu	
	Countrie	Countries of origin exclude top income		Countri	Countries of origin		me quintile	
			qu					
In Temperature	3.381	4.975***	4.397*	5.918***	-1.668	-0.486	-0.930	0.072
	(2.380)	(1.823)	(2.577)	(1.897)	(1.890)	(1.828)	(1.491)	(1.321)
In Temp. interacted with poor		-17.172***		-18.079**		-12.923***		-12.000***
		(6.499)		(7.209)		(4.822)		(4.540)
In Precipitations	0.091	0.126	0.446	0.515	-0.320	-0.226	-0.724	-0.659
	(0.360)	(0.369)	(0.462)	(0.485)	(0.411)	(0.423)	(0.458)	(0.476)
In Prec. interacted with poor		-0.774		-1.062		-1.337		-0.889
		(1.703)		(1.711)		(1.385)		(1.401)
Country of origin Fixed Effects	yes	yes	yes	yes	yes	yes	yes	yes
Decade X Region effects	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Decade X Poor effects	yes	yes	yes	yes	yes	yes	yes	yes
Observations	458	458	462	462	458	458	462	462
R-squared	0.115	0.129	0.080	0.095	0.407	0.413	0.441	0.446
Number of countries of origin	115	115	116	116	115	115	116	116
Tem effect on poor countries		-12.197*		-12.161*		-13.410***		-11.928***

Note: Each column corresponds to a different Least Square estimated regression with fixed effects. In columns 1-4 we only include emigration to non-OECD destinations. In columns 5-8 we include emigrants to OECD destinations. Columns 1-2 and 5-6 use as a sample of poor/middle-income countries those non in OECD. Columns 3-4 and 7-8 instead use countries of origin in the bottom to the fourth quintiles in the per-capita GDP distribution. The weather station data are averaged using population weights. The standard errors are cluster by country of origin. *, **, *** indicate significance at the 10, 5 and 1% confidence level.

Table 6: Temperature and Emigration by destination distance

Poor and Middle-Income countries of origin included, years 1970-2000.

Dependent Variable= In(Emigration/population).	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	
		Close destination	ons (<1000 ki	<u>m)</u>	Distant destinations (> 1000 km)				
	Non	-OECD	Countries of origin		Non-OECD		Countries of origin exclude		
	Countries of origin		exclude top income		Countries of origin		top income quintile		
		quintile							
In Temperature	8.953**	10.452***	7.892*	9.424**	-0.247	1.442	1.406	3.042**	
	(3.505)	(3.242)	(4.185)	(3.736)	(2.057)	(1.530)	(1.861)	(1.346)	
In Temp. interacted with poor		-20.439		-22.198		-18.398**		-19.614**	
		(13.260)		(13.563)		(7.080)		(7.563)	
In Precipitations	-0.668	-0.409	-0.764	-0.497	-0.396	-0.296	-0.496	-0.392	
	(0.705)	(0.718)	(0.820)	(0.840)	(0.379)	(0.376)	(0.415)	(0.409)	
In Prec. interacted with poor		-4.050		-3.580		-1.540		-1.443	
		(3.250)		(3.316)		(1.566)		(1.577)	
Country of origin Fixed Effects	yes	yes	yes	yes	yes	yes	yes	yes	
Decade X Region effects	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	
Decade X Poor effects	yes	yes	yes	yes	yes	yes	yes	yes	
Observations	393	393	391	391	458	458	462	462	
R-squared	0.106	0.115	0.084	0.093	0.201	0.219	0.228	0.248	
Number of countries of origin	106	106	106	106	115	115	116	116	
Tem effect on poor countries		-9.987		-12.774		-16.956**		-16.572**	

Note: Each column corresponds to a different Least Square estimated regression with fixed effects. In columns 1-4 we only include emigration to close destinations (<1000 Km) destinations. In columns 5-8 we include emigrants to distant destinations (>1000 Km). Columns 1-2 and 5-6 use as a sample of poor/middle-income countries those non in OECD. Columns 3-4 and 7-8 instead use countries of origin in the bottom to the fourth quintiles in the per-capita GDP distribution. The weather station data are averaged using population weights. The standard errors are cluster by country of origin. *, **, *** indicate significance at the 10, 5 and 1% confidence level.

Table 7: Temperature and agriculture share in GDP years 1970-2000

	(1)	(2)	(3)	(4)
	Non-C	DECD	Countries of origir	n exclude top income
Dependent variable (VA in agriculture/GDP)	Countries	of origin	qu	iintile
Temperature	-1.435	-2.205*	-1.502	-2.656*
	(1.341)	(1.256)	(1.685)	(1.566)
Temperature X poor		3.846		5.295
		(3.857)		(4.282)
Precipitations	-0.598**	-0.672**	-0.607**	-0.681**
	(0.277)	(0.295)	(0.288)	(0.307)
Precipitation X poor		0.895		0.795
		(1.052)		(1.111)
Country of origin Fixed Effects	Yes	Yes	Yes	Yes
Decade X Region effects	Yes	Yes	Yes	Yes
Decade X Poor effects	Yes	Yes	Yes	Yes
Observations	325	325	317	317
R-squared	0.328	0.334	0.338	0.346
Number of countries of origin	94	94	90	90
Temperature effect in poor countries		1.641		2.639

Poor and Middle-Income countries of origin included, years 1970-2000.

Note: Each column corresponds to a different Least Square estimated regression with fixed effects. The sample of countries for columns 1-2 are all non-OECD countries. Columns 3-4 use a sample of poor/middle-income countries of origin in the bottom to the fourth quintiles in the per-capita GDP distribution. The weather station data are averaged using population weights. Data on the share of agriculture in GDP are from a World Development Indicator database. The standard errors are cluster by country of origin. *, **, *** indicate significance at the 10, 5 and 1% confidence level.

Table 8: Temperature and GDP per capita

Poor and Middle-Income countries of origin included, years 1970-2000.

Dependent Variable= In(GDP per capita)	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
		Non-OECD Cou	intries of orig	in	Countrie	s of origin exclu	ide top incom	e quintile
	Penn World Table source		WDI s	source	Penn World Table source		WDI source	
Ln(Temperature)	1.750**	2.283**	1.462*	1.909**	1.021	1.414**	1.106	1.492**
	(0.837)	(0.981)	(0.752)	(0.772)	(0.705)	(0.631)	(0.710)	(0.641)
Ln(Temperature) X poor		-5.910**		-5.147*		-4.785*		-4.747*
		(2.811)		(2.614)		(2.710)		(2.577)
In Precipitations	0.382**	0.427**	0.358**	0.383**	0.392**	0.439**	0.373**	0.401**
	(0.169)	(0.174)	(0.149)	(0.158)	(0.178)	(0.184)	(0.146)	(0.156)
In (Precipitations) X poor		-0.569		-0.499		-0.514		-0.443
		(0.512)		(0.498)		(0.539)		(0.488)
Country of origin Fixed Effects	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Decade X Region effects	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Decade X Poor effects	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Observations	446	446	386	386	450	450	392	392
R-squared	0.303	0.314	0.297	0.305	0.380	0.388	0.399	0.406
Number of countries of origin	113	113	107	107	114	114	109	109
Tem effect on poor countries		-3.626		-3.238		-3.370		-3.255

Note: Note: Each column corresponds to a different Least Square estimated regression with fixed effects. The sample for columns 1-4 are all non-OECD countries. Columns 5-8 use a sample of poor/middle-income countries of origin in the bottom to the fourth quintiles in the per-capita GDP distribution. The weather station data are averaged using population weights. Columns 1-2 and 5-6 use data on GDP from Penn World Table. Columns 3-4 and 7-8 instead use data on GDP from a World Development Indicator database. The standard errors are cluster by country of origin. *, **, *** indicate significance at the 10, 5 and 1% confidence level.

Table Appendix

Table A1: Precipitations and emigration

Poor and Middle-Income countries of origin included, years 1970-2000.

	(1)	(2)	(3)	(4)	(5)	(6)
Dependent Variable=						
Ln(Emigration/Population)	Population weig Countries	Population weights. Non-OECD Countries of origin		ts. Non-OECD s of origin	Population weights. Countries of orig exclude top income quintile	
In Precipitations	-0.399	-0.364	-0.070	-0.164	-0.485	-0.451
	(0.364)	(0.342)	(0.361)	(0.360)	(0.426)	(0.406)
In (Precipitations) X poor		-0.363		0.910		-0.305
		(2.067)		(1.690)		(2.097)
Country of origin Fixed Effects	Yes	Yes	Yes	Yes	Yes	Yes
Decade X Region effects	Yes	Yes	Yes	Yes	Yes	Yes
Decade X Poor effects	Yes	Yes	Yes	Yes	Yes	Yes
Observations	462	462	456	456	466	466
R-squared	0.162	0.162	0.154	0.155	0.175	0.175
Number of countries of origin	116	116	115	115	117	117
Predicted effect on poor countries		-0.727		0.746		-0.756

Note: Each column corresponds to a different Least Square estimated regression with fixed effects. The sample of countries for columns 1-4 are all non-OECD countries. In columns 1-2 and 5-6 the weather station data are averaged using population weights. Columns 5-6 use area as weight. Columns 7-8 use a sample of poor/middle-income countries of origin in the bottom to the fourth quintiles in the per-capita GDP distribution. The standard errors are cluster by country of origin. *, **, *** indicate significance at the 10, 5 and 1% confidence level.

Table A2: Temperature and emigration, decade differences

Dependent Variable= Δ In(Emigration rates)	(1)	(2)	(3)	(4)
			Countries of origin	exclude top income
	Non-OECD Co	untries of origin	qu	intile
Δ In(Temperature)	2.291*	2.911**	2.656**	3.408***
	(1.281)	(1.368)	(1.223)	(1.299)
Δ Ln(Precipitation)	-0.416		-0.361	
	(0.376)		(0.452)	
Δ ln(Temp) X poor		-6.302		-7.553
		(5.070)		(5.384)
Δ Ln(Precipitation) X poor		-0.648		-0.932
		(2.156)		(2.166)
Year X Area	yes	yes	yes	yes
Year X Poor	yes	yes	yes	yes
Observations	343	343	346	346
R-squared	0.156	0.159	0.175	0.179
Number of countries of origin	343	343	346	346
Country fixed effects	No	No	No	No
Δ ln(Temp) effect on poor countries		-3.391		-4.145

Poor and Middle-Income countries of origin included, years 1970-2000.

Note: Each column corresponds to a different estimated regression. The sample for columns 1-2 are all non-OECD countries. Columns 3-4 use a sample of poor/middle-income countries of origin in the bottom to the fourth quintiles in the per-capita GDP distribution. The units of observations are decade-differences for each country. Method of estimation is ordinary least squares. The weather station data are averaged using population weights. The standard errors are cluster by country of origin. *, **, *** indicate significance at the 10, 5 and 1% confidence level.

Table A3: Temperature and emigration. Long differences

Poor and Middle-Income countries of origin included, 1970-2000

	(1)	(2)	(3)	(4)
Dependent Variable= Δ In(Emigration rates)	O	CD	Countries of original	n exclude top income
	origin count	ries excluded	qı	lintile
Δ In(Temperature)	2.904	4.277*	2.450	3.637*
	(2.792)	(2.481)	(2.457)	(2.040)
Δ ln(Temp) X poor		-35.339***		-35.299***
		(8.906)		(8.676)
Δ In (Precipitations)	-0.269	-0.197	-0.524	-0.496
	(0.424)	(0.444)	(0.471)	(0.490)
Δ In (Precipitations) X poor		-3.649**		-3.281*
		(1.808)		(1.757)
Area Dummies	Yes	Yes	Yes	Yes
Poor Dummy	Yes	Yes	Yes	Yes
Observations	114	114	115	115
R-squared	0.211	0.271	0.248	0.307
Tem effect on poor countries		-31.062***		-31.663***

Note: Each column corresponds to a different estimated regression. The sample for columns 1-2 are all non-OECD countries. Columns 3-4 use a sample of poor/middle-income countries of origin in the bottom to the fourth quintiles in the per-capita GDP distribution. The units of observations are the difference between the average temperatures in the last decade of our sample (1990-2000) and the average temperatures in the first decade (1970–1980). The weather station data are averaged using population weights. Method of estimation is ordinary least squares. *, **, *** indicate significance at the 10, 5 and 1% confidence level.

	(1)	(2)	(3)	(4)
	Dependent Varia Ln(Emigration ra	ıble tes)	Dependent variable: Urbanizatior Rates	
In (Temperature)	3.274*	3.778**	0.848**	0.972**
	(1.666)	(1.715)	(0.398)	(0.430)
ln (Temp) X Poor		-19.966**		-1.237
		(8.721)		(0.964)
ln (Temp) X SSA	-11.139**	-0.249	-1.325*	-0.649
	(5.438)	(6.663)	(0.744)	(1.077)
Ln(Precipitation)	-0.680*	-0.583*	0.017	0.017
	(0.348)	(0.313)	(0.042)	(0.042)
Ln(Precipitation) X (poor)		-2.683		-0.085
		(1.742)		(0.090)
Ln(Precipitation)X SSA	2.237*	3.114**	-0.149*	-0.104
	(1.274)	(1.391)	(0.076)	(0.079)
Country of origin Fixed Effects	yes	Yes	Yes	yes
Decade X Region effects	yes	Yes	Yes	yes
Decade X Poor effects	yes	Yes	Yes	yes
Observations	458	458	565	565
R-squared	0.203	0.221	0.731	0.734
Number of countries of origin	115	115	114	114
Tem effect on poor countries		-16.188*		-0.265

Table A4: Temperature and emigration/urbanization. Additional Interactions with SSA

Poor and Middle-Income countries of origin included, 1970-2000

Note: Each column corresponds to a different Least Square estimated regression with fixed effects. The sample includes non-OECD countries. Columns 1-2 use the natural logarithm of emigration rates as dependent variable. Columns 3-4 use urbanization rates as dependent variable. The weather station data are averaged using population weights. The standard errors are cluster by country of origin. *, **, *** indicate significance at the 10, 5 and 1% confidence level

Table A5: Control for Extreme Temperatures and Emigration

Poor and Middle-Income	countries (of oriain i	ncluded.	1970-2000
	countries	oj oligini i	neraaca;	10,0 2000

Dependent Variable= In(Emigration rates)	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)
	Population origin cour	weights. OECD tries excluded		Population exclude top	weights. Count income quintil	ries of origin e	Population origin count extremes (e	weights. OECD tries excluded. excluded colder	Only hotter extremes)
In (Temperature)	1.390	3.101*	3.099*	2.048	3.641***	3.661***	1.477	3.240**	2.884*
	(1.998)	(1.578)	(1.601)	(1.989)	(1.304)	(1.315)	(1.967)	(1.590)	(1.604)
Ln(Precipitation)	-0.265	-0.321	-0.321	-0.277	-0.341	-0.348	-0.335	-0.382	-0.385
	(0.375)	(0.369)	(0.370)	(0.446)	(0.443)	(0.443)	(0.363)	(0.356)	(0.357)
Extreme Temperature	0.106	0.102	0.103	0.084	0.082	0.074	0.026	0.020	0.065
	(0.068)	(0.068)	(0.074)	(0.066)	(0.065)	(0.072)	(0.085)	(0.084)	(0.093)
In (Temperature) X Poor		-18.338***	-18.336***		-18.442***	-18.466***		-18.566***	-17.475***
		(6.570)	(6.559)		(6.961)	(6.970)		(6.510)	(6.609)
Ln(Precipitation) X Poor		0.004	0.001		0.016	0.065		-0.067	-0.191
		(1.871)	(1.881)		(1.884)	(1.891)		(1.887)	(1.888)
Extreme Temperature X poor			-0.002			0.047			-0.278
			(0.176)			(0.176)			(0.236)
Country of origin Fixed Effects	yes	yes	yes	yes	yes	yes	yes	yes	yes
Decade X Region effects	yes	yes	yes	yes	yes	yes	yes	yes	yes
Decade X Poor effects	yes	yes	yes	yes	yes	yes	yes	yes	yes
Observations	448	448	448	456	456	456	448	448	448
R-squared	0.183	0.203	0.203	0.198	0.219	0.219	0.177	0.198	0.201
Number of countries of origin	112	112	112	114	114	114	112	112	112
Temperature effect on poor countries		-15.237**	-15.237**		-14.800**	-14.805**		-15.326**	-14.591**

Note: Each column corresponds to a different Least Square estimated regression with fixed effects. The sample for columns 1-3 and 7-9 are all non-OECD countries. Columns 4-6 use a sample of poor/middle-income countries of origin in the bottom to the fourth quintiles in the per-capita GDP distribution. The "Extreme temperature" variable is defined as the number of years in a decade in which the temperature was above or below two standard deviations of the 1960-2000 period mean for the country. In column 7-9 only the episodes of temperature above the average plus two standard deviations are included in the definition of 'Extreme temperature". The weather station data are averaged using population weights. The standard errors are cluster by country of origin. *, **, *** indicate significance at the 10, 5 and 1% confidence level.

Table A6: Control for natural Disasters and emigration

Poor and Middle-Income countries of origin included, 1970-2000

Dependent Variable= In(Emigration rates)	(1)	(2)	(3)	(4)	(5)	(6)
	Population origin cour	weights. OECD ntries excluded)	Population exclude top	weights. Count income quintil	ries of origin e
ln (Temperature)	1.561	3.414**	3.382*	2.404	4.153***	4.141***
	(1.787)	(1.695)	(1.709)	(1.641)	(1.244)	(1.251)
Ln(Precipitation)	-0.264	-0.196	-0.196	-0.322	-0.248	-0.247
	(0.354)	(0.329)	(0.330)	(0.423)	(0.397)	(0.397)
Natural Disasters	-0.007*	-0.005	-0.005	-0.005	-0.004	-0.004
	(0.004)	(0.004)	(0.004)	(0.004)	(0.005)	(0.005)
In (Temperature) X Poor		-19.310***	-19.368***		-19.661***	-19.693***
		(6.628)	(6.641)		(7.134)	(7.195)
Ln(Precipitation) X Poor		-1.314	-1.446		-1.254	-1.324
		(1.901)	(1.826)		(1.914)	(1.858)
Natural Disasters X poor			0.010			0.005
			(0.031)			(0.031)
Country of origin Fixed Effects	yes	yes	yes	yes	yes	yes
Decade X Region effects	yes	yes	yes	yes	yes	yes
Decade X Poor effects	yes	yes	yes	yes	yes	yes
Observations	458	458	458	462	462	462
R-squared	0.183	0.203	0.204	0.198	0.219	0.219
Number of countries of origin	115	115	115	116	116	116
Temperature effect on poor countries		-15.896**	-15.986**		-15.508**	-15.553**

Note: Each column corresponds to a different Least Square estimated regression with fixed effects. The sample for columns 1-3 are all non-OECD countries. Columns 4-6 use a sample of poor/middle-income countries of origin in the bottom to the fourth quintiles in the per-capita GDP distribution. The "natural Disasters" variable is defined as the number of times in a decade that a natural disaster occurred .The weather station data are averaged using population weights. The standard errors are cluster by country of origin. *, **, *** indicate significance at the 10, 5 and 1% confidence level.

Table A7: Temperature and emigration, Robustness check Temperature in levels

Dependent Variable= In(Emigration rates).	(1)	(2)	(3)	(4)	(5)	(6)
	Population w origin countr	eights. OECD ies excluded	Area weigh countrie	ts. OECD origin es excluded	Population weights exclude top in	. Countries of origin ncome quintile
Temperature	0.036	0.267*	-0.151	0.075	0.101	0.352**
	(0.161)	(0.155)	(0.150)	(0.166)	(0.166)	(0.157)
Precipitations	-0.030	-0.013	-0.025	-0.012	-0.037	-0.021
	(0.028)	(0.024)	(0.031)	(0.028)	(0.027)	(0.023)
Temperature X poor		-1.127***		-0.844**		-1.219***
		(0.336)		(0.343)		(0.352)
Precipitations X poor		-0.159		-0.120		-0.154
		(0.116)		(0.155)		(0.116)
Country of origin Fixed Effects	yes	yes	yes	yes	yes	yes
Decade X Region effects	yes	yes	yes	yes	yes	yes
Decade X Poor effects	yes	yes	yes	yes	yes	yes
Observations	458	458	452	452	462	462
R-squared	0.057	0.178	0.166	0.183	0.193	0.225
Number of countries of origin	115	115	114	114	116	116
Tem effect on poor countries		-0.860***		-0.7691**		-0.867***

Poor and Middle-Income countries of origin included, 1970-2000

Note: Each column corresponds to a different Least Square estimated regression with fixed effects. The sample of countries for columns 1-4 are all non-OECD countries. In columns 1-2 the weather station data are averaged using population weights. Columns 3-4 use area as weight. Columns 5-6 use a sample of poor/middle-income countries of origin in the bottom to the fourth quintiles in the per-capita GDP distribution. The standard errors are cluster by country of origin. *, **, *** indicate significance at the 10, 5 and 1% confidence level.

Figure 1: Temperature increase and Migration-Skill Thresholds

Illustration of the Theoretical model



Panel 1: Middle Income Country





Figure 2: Cumulated Changes in Emigration Rates

Selected countries at each decile of the distribution, 1970-2010



Figure 3: Cumulated Changes in Average Temperatures

Selected countries at each decile of the distribution, 1970-2000









Note: The graphs plot on the horizontal axis the natural logarithm of the average temperatures between 2000 and 1981 minus the natural logarithm of the average temperatures between 1960 and 1980. On the vertical axis the natural logarithm of the average emigration rates between 1990 and 2000 minus the average emigration rates between 1970 and 1980.

NOTE DI LAVORO DELLA FONDAZIONE ENI ENRICO MATTEI

Fondazione Eni Enrico Mattei Working Paper Series

Our Note di Lavoro are available on the Internet at the following addresses:

http://www.feem.it/getpage.aspx?id=73&sez=Publications&padre=20&tab=1 http://papers.ssrn.com/sol3/JELJOUR_Results.cfm?form_name=journalbrowse&journal_id=266659

http://ideas.repec.org/s/fem/femwpa.html

http://www.econis.eu/LNG=EN/FAM?PPN=505954494

http://ageconsearch.umn.edu/handle/35978

http://www.bepress.com/feem/

NOTE DI LAVORO PUBLISHED IN 2015

ERM	1.2015	Elena Verdolini, Laura Diaz Anadon, Jiaqi Lu and Gregory F. Nemet: <u>The Effects of Expert Selection</u> ,
CCCD	0.0015	Elicitation Design, and R&D Assumptions on Experts' Estimates of the Future Costs of Photovoltaics
CCSD	2.2015	James Lennox and Ramiro Parrado: <u>Capital-embodied Technologies in CGE Models</u>
CCSD	3.2015	Claire Gavard and Djamel Kirat: <u>Flexibility in the Market for International Carbon Credits and Price</u>
CCCD	4 2015	Dynamics Difference with European Allowances
CCSD	4.2015	Claire Gavard: <u>Carbon Price and Wind Power Support in Denmark</u>
CCSD	5.2015	Wook Near Comerce Deliver and the caving Ended De Clan and Ender Kriegier: <u>implications of</u>
	6 2015	weak Near-term Contracts on Long-term Wildgatton ratinways
CC3D	0.2013	Stackalberg Competition
CCSD	7 2015	C. Dianicia Bérra Blanca and Thomas Thalar: Water Flows in the Economy. An Input-output Framework to
CCSD	7.2013	Access Water Productivity in the Castile and León Region (Spain)
CCSD	8 2015	Carlos M. Gómez and C. Dionisio Pérez-Blanco: Simple Myths and Basic Maths about Greening Irrigation
CCSD	9 2015	Elori loss Benedetto Rugani Samer Rege Enrico Benetto Laurent Drouet Dan Zachary and Tom Hass
CCOD	5.2010	Integrated Environmental Assessment of Future Energy Scenarios Based on Economic Fauilibrium Models
ERM	10.2015	Beatriz Martínez and Hipòlit Torró: European Natural Gas Seasonal Effects on Futures Hedging
CCSD	11.2015	Inge van den Bijgaart: The Unilateral Implementation of a Sustainable Growth Path with Directed Technical
		Change
CCSD	12.2015	Emanuele Massetti, Robert Mendelsohn and Shun Chonabayashi: Using Degree Days to Value Farmland
CCSD	13.2015	Stergios Athanassoglou: Revisiting Worst-case DEA for Composite Indicators
CCSD	14.2015	Francesco Silvestri and Stefano Ghinoi : Municipal Waste Selection and Disposal: Evidences from Lombardy
CCSD	15.2015	Loïc Berger: The Impact of Ambiguity Prudence on Insurance and Prevention
CCSD	16.2015	Vladimir Otrachshenko and Francesco Bosello: Identifying the Link Between Coastal Tourism and Marine
		Ecosystems in the Baltic, North Sea, and Mediterranean Countries
ERM	17.2015	Charles F. Mason, Lucija A. Muehlenbachs and Sheila M. Olmstead: The Economics of Shale Gas
		<u>Development</u>
ERM	18.2015	Anna Alberini and Charles Towe: Information v. Energy Efficiency Incentives: Evidence from Residential
		Electricity Consumption in Maryland
CCSD	19.2015	ZhongXiang Zhang: Crossing the River by Feeling the Stones: The Case of Carbon Trading in China
CCSD	20.2015	Petterson Molina Vale: The Conservation versus Production Trade-off: Does Livestock Intensification
		Increase Deforestation? The Case of the Brazilian Amazon
CCSD	21.2015	Valentina Bosetti, Melanie Heugues and Alessandro Tavoni: <u>Luring Others into Climate Action: Coalition</u>
CCCD	22 224 5	Formation Games with Threshold and Spillover Effects
CCSD	22.2015	Francesco Bosello, Elisa Delpiazzo, and Fabio Eboli: <u>Macro-economic Impact Assessment of Future Changes</u>
CCCD	22 2015	In European Marine Ecosystem Services
CCSD	23.2015	Maryse Labriet, Laurent Drouet, Marc Vielle, Richard Louiou, Amit Kanudia and Alain Haurie: <u>Assessment or</u>
CCSD	24 2015	the Effectiveness of Global Climate Policies Using Coupled Bottom-up and Top-down Models
CCSD	24.2013	Technological Programs
CCSD	25 2015	Renjamin Michallet, Giusenne Lucio Gaeta and François Facchini: Greening Un or Not? The Determinants
CCJD	25.2015	Political Parties' Environmental Concern: An Empirical Analysis Resed on Furonean Data (1970-2008)
CCSD	26 2015	Daniel Bodansky, Seth Hoedl, Gilbert Metcalf and Robert Stavins: Eacilitating Linkage of Heterogeneous
CCDD	20.2015	Regional National and Sub-National Climate Policies Through a Future International Agreement
CCSD	27,2015	Giannis Vardas and Anastasios Xepapadeas: Time Scale Externalities and the Management of Renewable
		Resources
CCSD	28.2015	Todd D. Gerarden, Richard G. Newell, Robert N. Stavins and Robert C. Stowe: An Assessment of the
		Energy-Efficiency Gap and Its Implications for Climate Change Policy
CCSD	29.2015	Cristina Cattaneo and Emanuele Massetti: <u>Migration and Climate Change in Rural Africa</u>
ERM	30.2015	Simone Tagliapietra: The Future of Renewable Energy in the Mediterranean. Translating Potential into
		Reality
CCSD	31.2015	Jan Siegmeier, Linus Mattauch, Max Franks, David Klenert, Anselm Schultes and Ottmar Edenhofer: <u>A Public</u>
		Finance Perspective on Climate Policy: Six Interactions That May Enhance Welfare
CCSD	32.2015	Reyer Gerlagh, Inge van den Bijgaart, Hans Nijland and Thomas Michielsen: Fiscal Policy and CO2 Emissions
		of New Passenger Cars in the EU
CCSD	33.2015	Marie-Laure Nauleau, Louis-Gaëtan Giraudet and Philippe Quirion: Energy Efficiency Policy with Price-
		<u>quality Discrimination</u>

CCSD	34.2015	Eftichios S. Sartzetakis, Anastasios Xepapadeas and Athanasios Yannacopoulos: <u>Regulating the</u>
		Environmental Consequences of Preferences for Social Status within an Evolutionary Framework
CCSD	35.2015	Todd D. Gerarden, Richard G. Newell and Robert N. Stavins: <u>Assessing the Energy-efficiency Gap</u>
CCSD	36.2015	Lorenza Campagnolo and Fabio Eboli: <u>Implications of the 2030 EU Resource Efficiency Target on</u>
		Sustainable Development
CCSD	37.2015	Max Franks. Ottmar Edenhofer and Kai Lessmann: Why Finance Ministers Favor Carbon Taxes. Even if They
		Do not Take Climate Change into Account
CCSD	38 2015	ZhongXiang Zhang: Carbon Emissions Trading in China: The Evolution from Pilots to a Nationwide Scheme
CCSD	30.2015	David Carsis Loon Weather and Income Lassons from the Main European Paging
CCSD	10 2015	David Garcia-Leon. <u>Weather and monie. Lessons from the Affrain European Regions</u>
CCSD	40.2015	Jaroslav Myslak and C. D. Perez-Bianco: Partnersnips for Antoroable and Equitable Disaster Insurance
CCSD	41.2015	S. Surminski, J.C.J.H. Aerts, W.J.W. Botzen, P. Hudson, J. Mysiak and C. D. Perez-Blanco: <u>Reflections on the</u>
		<u>Current Debate on How to Link Flood Insurance and Disaster Risk Reduction in the European Union</u>
CCSD	42.2015	Erin Baker, Olaitan Olaleye and Lara Aleluia Reis: <u>Decision Frameworks and the Investment in R&D</u>
CCSD	43.2015	C. D. Pérez-Blanco and C. M. Gómez: <u>Revealing the Willingness to Pay for Income Insurance in Agriculture</u>
CCSD	44.2015	Banchongsan Charoensook: On the Interaction between Player Heterogeneity and Partner Heterogeneity in
		Two-way Flow Strict Nash Networks
CCSD	45.2015	Frin Baker, Valentina Bosetti, Laura Diaz Anadon, Max Henrion and Lara Aleluia Reis: Euture Costs of Key
CCDD	10.2010	Low-Carbon Energy Technologies: Harmonization and Aggregation of Energy Technology Expert Elicitation
		Data
CCCD	46 2015	
CCSD	46.2015	Sushanta Kumar Mahapatra and Keshab Chandra Ratha: <u>Sovereign States and Surging Water: Brahmaputra</u>
		River between China and India
CCSD	47.2015	Thomas Longden: <u>CO2</u> Intensity and the Importance of Country Level Differences: An Analysis of the
		Relationship Between per Capita Emissions and Population Density
CCSD	48.2015	Jussi Lintunen and Olli-Pekka Kuusela: Optimal Management of Markets for Bankable Emission Permits
CCSD	49.2015	Johannes Emmerling: Uncertainty and Natural Resources - Prudence Facing Doomsday
FRM	50 2015	Manfred Hafner and Simone Taglianietra: Turkish Stream: What Strategy for Europe?
ERM	51 2015	Thomas Sattick land Videsboard and Daniel Scholten: Can Ell's Decarbonization Agenda Break the State-
	51.2015	Comments Sattich, inga Testsbolid and Daniel Schöten. Car LO's Decarbonisation Agenda Dreak the State-
5514	50 004 5	Company Axis in the Power Sector?
ERM	52.2015	Alessandro Cologni, Elisa Scarpa and Francesco Giuseppe Sitzia: <u>Big Fish: Oil Markets and Speculation</u>
CCSD	53.2015	Joosung Lee: <u>Multilateral Bargaining in Networks: On the Prevalence of Inefficiencies</u>
CCSD	54.2015	P. Jean-Jacques Herings: <u>Equilibrium and Matching under Price Controls</u>
CCSD	55.2015	Nicole Tabasso: Diffusion of Multiple Information: On Information Resilience and the Power of Segregation
CCSD	56.2015	Diego Cerdeiro, Marcin Dziubinski and Sanjeev Goyal: Contagion Risk and Network Design
CCSD	57.2015	Yann Rébillé and Lionel Richefort: Networks of Many Public Goods with Non-Linear Best Replies
CCSD	58,2015	Achim Hagen and Klaus Eisenack: International Environmental Agreements with Asymmetric Countries:
0000	00.20.0	Climate Clubs vs. Global Cooperation
CCSD	50 2015	<u>Cimitate Guids vs. Giobal and Vincent Vennetalbaseb. Constitutions and Sasial Networks</u>
CCSD	59.2015	And invalueon, Nils Roeni and Vincent Vanneteibosch: <u>Constitutions and social Networks</u>
CCSD	60.2015	Adam N. Walker, Hans-Peter Welkard and Andries Kichter: <u>The Rise and Fall of the Great Fish Pact under</u>
		Endogenous Risk of Stock Collapse
CCSD	61.2015	Fabio Grazi and Henri Waisman: <u>Agglomeration, Urban Growth and Infrastructure in Global Climate Policy:</u>
		<u>A Dynamic CGE Approach</u>
CCSD	62.2015	Elorri Igos, Benedetto Rugani, Sameer Rege, Enrico Benetto, Laurent Drouet and Dan Zachary: Combination
		of Equilibrium Models and Hybrid Life Cycle-Input-Output Analysis to Predict the Environmental Impacts of
		Energy Policy Scenarios
CCSD	63 2015	Delayane B. Diaz: Estimating Clobal Damages from Sea Level Rise with the Coastal Impact and Adaptation
CCSD	00.2010	Model (CIAM)
CCCD	64 2015	<u>Midder (Chawr)</u>
CCSD	64.2015	Delavane B. Diaz: Integrated Assessment of Climate Catastrophes with Endogenous Uncertainty: Does the
		Risk of Ice Sheet Collapse Justify Precautionary Mitigation?
CCSD	65.2015	Jan Witajewski-Baltvilks, Elena Verdolini and Massimo Tavoni: <u>Bending The Learning Curve</u>
CCSD	66.2015	W. A. Brock and A. Xepapadeas: Modeling Coupled Climate, Ecosystems, and Economic Systems
CCSD	67.2015	Ricardo Nieva: The Coalitional Nash Bargaining Solution with Simultaneous Payoff Demands
CCSD	68.2015	Olivier Durand-Lasserve, Lorenza Campagnolo, Jean Chateau and Rob Dellink: Modelling of Distributional
		Impacts of Energy Subsidy Reforms: an Illustration with Indonesia
CCSD	69 2015	Simon Levin and Anastasios Xenanadeas: Transhoundary Canital and Pollution Flows and the Emergence of
CCSD	09.2013	Decise al Inequalities
CCCD	70 2015	Regionar meduances
CCSD	70.2015	Jaroslav Mysiak, Swenja Surminski, Annegret Thieken, Reinhard Mechler and Jeroen Aerts: Sendai Framework
		for Disaster Risk Reduction – Success or Warning Sign for Paris?
CCSD	71.2015	Massimo Tavoni and Detlef van Vuuren: <u>Regional Carbon Budgets: Do They Matter for Climate Policy?</u>
CCSD	72.2015	Francesco Vona, Giovanni Marin, Davide Consoli and David Popp: <u>Green Skills</u>
CCSD	73.2015	Luca Lambertini, Joanna Poyago-Theotoky and Alessandro Tampieri: <u>Cournot Competition and "Green"</u>
		Innovation: An Inverted-U Relationship
FS	74.2015	Michele Raitano and Francesco Vona: From the Cradle to the Grave: the Effect of Family Background on the
-		Career Path of Italian Men
FS	75 2015	Davide Carbonai and Carlo Drago: Positive Freedom in Networked Capitalism: An Empirical Analysis
	76 2015	Wai lin and Zhang Viang Zhangi Lavalling the Dlaving Eigld. On the Missing Data of Maturals Estimation
CCSD	70.2015	vver jin and zhongstiang zhang. Levening the Playing Field. On the Wissing Kole of Network Externality in
		Designing Kenewable Energy Technology Deployment Policies
ERM	//.2015	Niaz Bashiri Behmiri and Matteo Manera: <u>The Role of Outliers and Oil Price Shocks on Volatility of Metal</u>
		Prices
CCSD	78.2015	Jan Witajewski-Baltvilks, Elena Verdolini and Massimo Tavoni: Directed Technological Change and Energy
		Efficiency Improvements

ES	79.2015	David Cuberes and Rafael González-Val: The Effect of the Spanish Reconquest on Iberian Cities
CCSD	80.2015	Isabella Alloisio, Alessandro Antimiani, Simone Borghesi, Enrica De Cian, Maria Gaeta, Chiara Martini,
		Ramiro Parrado, Maria Cristina Tommasino, Elena Verdolini and Maria Rosa Virdis: <u>Pathways to Deep</u>
		Carbonization in Italy
CCSD	81.2015	Yonky Indrajaya, Edwin van der Werf, Hans-Peter Weikard, Frits Mohren and Ekko C. van Ierland: <u>The</u>
		Potential of REDD+ for Carbon Sequestration in Tropical Forests: Supply Curves for carbon storage for
		<u>Kalimantan, Indonesia</u>
ES	82.2015	Carlo Drago, Roberto Ricciuti, Paolo Santella: <u>An Attempt to Disperse the Italian Interlocking Directorship</u>
		Network: Analyzing the Effects of the 2011 Reform
CCSD	83.2015	Joseph E. Aldy: <u>Policy Surveillance in the G-20 Fossil Fuel Subsidies Agreement: Lessons for Climate Policy</u>
CCSD	84.2015	Milan Ščasný, Emanuele Massetti, Jan Melichar and Samuel Carrara: Quantifying the Ancillary Benefits of the
		Representative Concentration Pathways on Air Quality in Europe
CCSD	85.2015	Frédéric Branger and Misato Sato: Solving the Clinker Dilemma with Hybrid Output-based Allocation
ERM	86.2015	Manfred Hafner and Simone Tagliapietra: <u>The Role of Natural Gas in the EU Decarbonisation Path</u>
CCSD	87.2015	Cristina Cattaneo and Giovanni Peri: The Migration Response to Increasing Temperatures